

Temi di discussione

(Working Papers)

Labor market spillovers of a large plant opening. Evidence from the oil industry

by Matteo Alpino, Irene Di Marzio, Maurizio Lozzi and Vincenzo Mariani





Temi di discussione

(Working Papers)

Labor market spillovers of a large plant opening. Evidence from the oil industry

by Matteo Alpino, Irene Di Marzio, Maurizio Lozzi and Vincenzo Mariani

Number 1386 - October 2022

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Antonio Di Cesare, Raffaela Giordano, Monica Andini, Marco Bottone, Lorenzo Braccini, Luca Citino, Valerio Della Corte, Lucia Esposito, Danilo Liberati, Salvatore Lo Bello, Alessandro Moro, Tommaso Orlando, Claudia Pacella, Fabio Piersanti, Dario Ruzzi, Marco Savegnago, Stefania Villa. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

LABOR MARKET SPILLOVERS OF A LARGE PLANT OPENING. EVIDENCE FROM THE OIL INDUSTRY

by Matteo Alpino*, Irene Di Marzio**, Maurizio Lozzi** and Vincenzo Mariani**

Abstract

We study the labor market spillovers of the opening of a large oil extracting facility in a peripheral area of the South of Italy, focusing on the time when the physical investment was made, before production took off and royalties started flowing in. Using firm-level administrative data covering the universe of non-farm businesses in a difference-indifferences design, we find large positive effects on firm-level employment and positive, albeit smaller, effects on wages per employee. Within-firm employment gains are geographically localized and are concentrated in the upstream sector (not only in manufacturing) and among the largest firms; however the number of firms declines. Aggregate results obtained using census data show that the overall employment effect in the area is slightly positive.

JEL Classification: J21, L71, O13, R11. Keywords: oil, local labor markets, plant opening, natural resource DOI: 10.32057/0.TD.2022.1386

Contents

| 1. Introduction | 5 |
|---|----|
| 2. Literature review | 8 |
| 3. Background | 10 |
| 4. Theoretical framework | 13 |
| 5. Data | 15 |
| 6. Firm level analysis | 17 |
| 6.1 Identification | 21 |
| 6.2 Estimated effects on the firm-level number of employees | 23 |
| 6.3 Estimated effects on the firm-level average wage | 35 |
| 6.4 The persistence of the effects | 37 |
| 7. Aggregate analysis | 38 |
| 7.1 Firms demographic | 39 |
| 7.2 Aggregate labor market effects | 40 |
| 8. Conclusion | 44 |
| References | 45 |
| Appendix A | 51 |
| Appendix B | 57 |

^{*} Bank of Italy, Directorate General for Economics, Statistics and Research; **Bank of Italy, Economic Analysis and Research Unit, Bari Branch.

1 Introduction¹

The opening of a large industrial plant often leads to an increase in local employment above and beyond the workers directly employed on the new site (Greenstone and Moretti, 2003; Monte et al., 2018); similar effects are found for the establishment of a large extraction facility (Aragón and Rud, 2013). In both cases employment gains are usually concentrated among incumbent firms that are geographically and economically close (i.e. in a buyer/supplier relationship or relying on similar technologies and workers) to the new entrant (Greenstone et al., 2010; Allcott and Keniston, 2018). Less is known on how local firms of different size and productivity are affected by the opening, although it seems reasonable that larger and more productive incumbents should be better equipped at accommodating the surge in demand prompted by the new entrant, pretty much in the same way as this type of firms is better equipped at accessing export markets and thus reap benefits from trade integration (Bernard et al., 2007; Melitz, 2003).

In this paper we provide new evidence on these issues. We focus on the opening of an oil extraction center in a remote area of the South of Italy (Val d'Agri, Basilicata) that hosts the largest oil reserves in Europe and we study its consequences on local firms in terms of employment and payroll. The richness of our administrative data, covering the universe of non-farm private businesses in the period since 1990, allows us to estimate heterogeneous effects in terms of sector, exposure to the oil industry, firm size, workforce composition and geographic location. Availability of data on construction and services is important, because most previous studies focused only on manufacturing, while a new industrial plant might generate demand also in other sectors. Notably, we study the time when the oil industry was first established in the area, before extraction and royalties took off; our estimates thus only refer to the initial physical capital investment (cumulatively 15 per cent of the annual regional GDP).

¹For comments we are grateful to Antonio Accetturo, Giuseppe Albanese, Gaetano Basso, Luca Citino, Domenico Depalo, Andreas Kotsadam, Halvor Mehlum, Litterio Mirenda, Sauro Mocetti, Filippo Scoccianti, Luca Sessa and Wessel Vermeulen. We thank Roberto Iacono for sharing data on oil production for early years, Luca Citino and Andrea Linarello for their help with the INPS data and Mariangela Grosso and Vittoria Roppo for valuable research assistance. All errors are our own.

From an econometric standpoint, we rely on difference-in-differences and eventstudy designs that compare firms located close to the plant to those further away around the time of the establishment of the oil industry; a similar strategy is used by Aragón and Rud (2013) to study the impact of a large gold mine in Peru on households.

We find that the opening had positive effects on employment at the firm-level: on average 1 more employee per firm, corresponding to a 30 per cent increase. Interestingly, the effects are much larger both in absolute and relative terms the larger the initial size: firms with more than 50 employees increase employment by more than 50 per cent, firms with less than 5 employees by only 2 per cent. Moreover, the share of small firms (up to 5 employees) in the local economy declines. In line with previous evidence, the estimated effects are very localized, and dissipate for businesses located more than 30 minutes away from the oil fields. The increase in employment concerns all economic sectors, but it is particularly strong in industries that sell a large share of their output to oil & gas companies. Remarkably, these upstream firms are not only manufacturers, but also belong to services.

We also explore the aggregate employment effect of the opening. To do so, we use data at the municipal level from the population census, that contains also information on self-employed, public employees and farmers. Total employment rate increases by 1 percentage point more in the oil commuting zone than in the surrounding local labor markets, and the effect is entirely driven by female employment. Thus the large firm-level increase does not translate into a sizable aggregate effect, likely because the number of private firms (especially large ones) is small and declining.

Taken together, our two pieces of evidence suggest that the opening triggered a reallocation of labor towards companies that were able to accommodate the demand generated by the new plant thanks to their geographical position, integration in the oil & gas value chain, or initial size. The finding that *size* complementarities matter is probably our most innovative contribution to the existing literature and might have interesting implications. Since large firms are on average more productive (Berlingieri et al., 2018; Brandolini and Bugamelli, 2009; Calligaris et al., 2016; Bugamelli et al.,

2018), our results might imply that the opening of a large industrial establishment will increase total productivity by favoring a reallocation of labor towards more productive companies. Unfortunately, we are unable to test directly this effect due to lack of firm-level balance sheet data for our sample period, which would allow an estimation of firms' productivity.

Our findings are informative for policymakers who must decide whether to favor or limit the extraction of natural resources or to subsidize or discourage the opening of large capital-intensive plants. From a policy perspective, it is important to know what are the likely winners and losers of these demand shocks, and to highlight that employment gains are not evenly distributed across firms and sectors. In fact, many placed-based policies favor the settlement of large firms in lagging-behind areas but, in light of our findings, these interventions might be scarcely effective if the incumbent local firms are too small or economically distant from the new entrant.²

An important caveat is that our study is limited to the labor market effects of the initial physical investment to set up the extractive industry, thus our evidence must be combined with estimates of the health (e.g. due to accidents), environmental (e.g. due to pollution) and fiscal (e.g. due to royalties) externalities in order to carry on a complete cost-benefit assessment of the presence of the oil industry, which is beyond the scope of this work.

The rest of the paper is organized as follows: section 2 illustrates how our work fits into different strands of the literature; section 3 provides background information about oil extraction and the economic structure of Val d'Agri; section 4 sketches a simple conceptual framework to guide the empirical analysis; section 5 describes the data; section 6 reports the analysis at the firm-level, and section 7 the analysis at the aggregate level; finally, section 8 concludes.

 $^{^{2}}$ Accetturo and de Blasio (2019) review the effect of these policies in Italy for the last 25 years, finding scarce support in favor of their effectiveness.

2 Literature review

In line with Greenstone et al. (2010) and Monte et al. (2018), we investigate the causal effects of the establishment of a large industrial plant on the surrounding firms. Greenstone et al. (2010) find that the total factor productivity of incumbent plants is considerably higher in U.S. counties that attracted a large manufacturing plant relative to counties that did not experience this shock. Using the same data, Greenstone and Moretti (2003) and Monte et al. (2018) find that the opening of a large manufacturing plant increases the local employment and the wage bill, especially in counties whose local labor market has a higher degree of openness in terms of commuting. Contrary to these papers, we do not study a manufacturing plant but an extractive facility, whose location does not depend on the attractiveness of the local economy, but rather on the presence of natural resources.

Given our focus on the extractive industry, we also contribute to the strand of the literature concerned with estimating the local impact of natural resource discovery and exploitation on labor market outcomes (Van Der Ploeg and Poelhekke (2017), Cust and Poelhekke (2015), Aragón et al. (2015) and Marchand and Weber (2018)). This literature, building on previous research at the country level (see Van der Ploeg (2011) for a review), has more recently taken advantage of novel disaggreggated data to model natural resource extraction as a source of local demand shocks. The empirical evidence overwhelmingly shows that resource extraction increases both employment and earnings at the regional level in advanced countries (Black et al., 2005; Michaels, 2010; Marchand, 2012; Aragón and Rud, 2013; Basso, 2017). Munasib and Rickman (2015) add that the intensity of positive spillovers to non-extractive industries depends positively on the degree of economic isolation of the extraction area. In particular, we share similarities with Pelzl and Poelhekke (2020), who analyze the local effect of natural resource extraction on a panel of manufacturing plants in Indonesia and find that labor-intensive booms induce crowding out of employment in the non-mining sectors, while capital-intensive mining booms do not affect local wages but increase employment. Finally, the focus on the investment period differentiates us from most other studies in the natural resource literature, that usually exploit for identification the temporal variation in resource prices (Basso, 2017) or in national employment in the extractive sector (Allcott and Keniston, 2018).³

As we study the establishment of a capital-intensive industry in a depressed area of the South of Italy, we also relate to the extensive literature on place-based policies, particularly to the strand focusing on the evaluation of industrial policies and physical capital investments⁴. The establishment of the oil industry in Basilicata was not a consequence of an explicit policy deployed to develop a lagging-behind area, even if, during most of the considered period, ENI was a State-owned company.

Since our analysis pays particular attention to treatment effect heterogeneity with respect to firm characteristics, we contribute to the debate on how demand shocks affect firm selection, labor reallocation and agglomeration spillovers. In analogy with the existing empirical evidence, we study how treatment effects vary with both spatial and economic distance from the extractive facilities. Regarding the former, we relate to the literature recently reviewed by Rosenthal and Strange (2020), which highlights that agglomeration economies are highly non-linear over space and decline with distance. As for the latter, we follow the approach in Allcott and Keniston (2018) and examine how the treatment effect varies with respect to the firms' degree of upstreamness to the oil industry. Our results that firms which are economically closer to the entrant plant are more impacted by the local shock is in line with both theoretical predictions and a growing body of empirical evidence studying how shocks propagate through networks of firms (Greenstone et al., 2010; Giroud and Mueller, 2019; Siegloch et al., 2021). Furthermore, we investigate heterogeneity in terms of firm size, finding that the treatment alters firms composition. This result speaks to the literature in international trade on the heterogeneous effects of economic integration and predicts that this shock leads to inter-firm reallocation towards more productive firms within the same industry

 $^{^{3}}$ One exception is Carrington (1996) who studies the labor market consequences of the construction of the trans-Alaska oil pipeline.

 $^{^{4}}$ See Kline and Moretti (2014), Neumark and Simpson (2015), Duranton and Venables (2018) for detailed reviews of the literature and Criscuolo et al. (2019); Siegloch et al. (2021) for particularly relevant recent contributions

(Melitz, 2003; Melitz and Ottaviano, 2008).

3 Background

Location The present work focuses on Val d'Agri (the valley of the Agri river), a small area that hosts the richest on-shore oil field in Europe. In the period 2004-18 the oil extractions from the area accounted for 99 per cent of the regional oil production and 70 per cent of the national oil production. The valley is surrounded by mountains (see Figure 1) and, according to the classification proposed by the Agenzia per la Coesione territoriale (2014), its municipalities are classified either as peripheral or ultra-peripheral internal zones, that is, areas far from the main centers of provision of public services (e.g. secondary and tertiary schools, hospitals, high-speed railways). The oil plant is located in the valley floor, in the municipality of Viggiano.

Val d'Agri is part of Basilicata, a small region (0.6 millions inhabitants) in the South of Italy.⁵ Similarly to other regions in the South, Basilicata has a lower GDP per capita than the Italian average (approximately 12,000 euros per capita, or 70 per cent of the Italian average in 1995), and a lower employment rate (38 per cent versus 45 in Italy in 1995, as a share of population with 15 years or older).

History The discovery of oil in Val d'Agri dates back to 1926. However, at the time extraction was difficult due to the rugged terrain and the depth of the reservoirs, so the operations stopped after some drilling between 1936 and 1945. In the 1970s exploration resumed, driven by technological progress and by the rise in oil price. In 1990 the industrial development of the oil sector began: the Ministry of Industry granted to the oil company ENI three new mining concessions (later merged under the name "Val d'Agri"), covering approximately 660 squared kilometers.⁶ At that time, ENI was a state-owned company, but was privatized in 1995, with the Italian treasury keeping control via a minority share. In 1996 ENI built the pilot treatment

⁵Basilicata is a NUTS 2 administrative region.

⁶The mining concession is a portion of land with well defined boundaries. It does not necessarily correspond to an administrative unit, and it can cut across administrative boundaries.



Figure 1: Geography of Val d'Agri (left) and its location in the South of Italy (right)

Source: Open Street Maps. Scale: 1:52,771 in the left panel; 1:1,678,047 in the right panel. Note: in the left panel, the red circle marks the location of the oil treatment center. Green areas are covered in forest. The grey regional border with Campania is visible in the south-west corner. In the right panel, the black square identifies Val d'Agri.

center in Val d'Agri, in the Viggiano municipality. The plant was operational since April and had a capacity of 7,500 barrels per day. Later that year, ENI requested the authorization to expand the plant, which was granted in 1999. At the end of 1998 ENI signed an exploitation agreement with the regional government, which included transfers and investments for 70 millions euro in 10 years. The new expanded plant was operational since 2001, with an additional treatment line added in 2002, and the final one in 2004. ENI also built a pipeline to carry the oil to the refinery in Taranto (150 kilometers away), operational since 2001. The new infrastructures increased the plant capacity by 14 times relative to the original pilot center. The overall investment by ENI amounted to roughly 1.5 billions euro (approximately 15 per cent of average regional annual GDP) according to Bank of Italy (2001). This magnitude roughly corresponds to the figures reported in the ENI balance sheet, where the attached documentation in the years between 1997 and 2000 reports the material immobilization in machinery and equipment specifically aimed at Val d'Agri; the scaling up of the investment over time is visible in Table 1.

| Year | Material investment | Regional GDP | Material investment |
|-------|---------------------|--------------------|-------------------------|
| | (millions of euro) | (millions of euro) | (share of regional GDP) |
| 1997 | 175.6 | 7,729.8 | 2.3 |
| 1998 | 300.6 | 8,088.5 | 3.7 |
| 1999 | 353 | 8,579.2 | 4.1 |
| 2000 | 475 | 8,825.4 | 5.4 |
| Total | 1,304.2 | | 15.7 |

Table 1: Ongoing material immobilization in machinery and equipment by ENI in Val d'Agri

Source: ENI for data on material investments and Istat for data on regional (NUTS2) GDP. Note: the table reports only figures for the years between 1990 and 2001 where the note to the balance sheet reports explicitly the amount of material investments aimed at the oil infrastructures in Val d'Agri. Last line of the table reports the share of regional GDP calculated as the ratio of overall material investment to average regional GDP between years 1997-2000.

Oil production closely tracked the expansion of the treatment capacity of the plant in Viggiano (see figure 2). Through the 1990s production raised steadily, but remained at relatively low levels (0.5 millions tons in 1999). When the expanded plant started its operations, production took off (130 per cent annual growth in 2002). After some further increases in the following years, production stabilized at more than 4 millions annual tons in 2005.



Figure 2: Oil production and capacity in Val d'Agri

Source: Ministry of Development and ENI. Note: thousands of tons. Vertical lines correspond to major expansions of the production capacity: the construction of the pilot treatment center, its expansion, and the addiction of two new production lines.

Royalties and oil price Since 1997 oil production on-shore is subject to a 7 per cent tax rate calculated on its market value, and its revenues accrue mostly to the regional government⁷. Between 1990 and 2001, the oil price was fairly stable around 20 US dollars per barrel; the low price combined with the low production made the revenues from royalties very limited in this period⁸. Although the revenues are not earmarked, the regional government used its share to finance a development plan that targeted thirty municipalities; the first cycle of the program took place in the period 2003-2008, and had a budget of 350 millions euro. At the end of 2008, only 19 per cent had been effectively spent (Bank of Italy, 2009).

Previous research In spite of the fact that Val d'Agri holds the largest onshore oil field in Europe, the evidence on the effects of its exploitation on the local economy is scant. Percoco (2012) finds a small positive effect on the number of active firms in the area where the development plan was active, using a geographic RDD at the municipality level. Iacono (2016) uses the synthetic control method at the regional (NUTS2) level and finds no effect on real per capita GDP, employment and investment. Descriptive works by Bubbico (2016, 2019) argue that the impact on local employment is likely to be very small due to the capital-intensive nature of oil industry and to the weakness of local institutions. On the contrary, company reports argue that the local employment effects are significant (ENI, 2012).

4 Theoretical framework

Several recent contributions have adapted the spatial equilibrium framework by Moretti (2010) and Moretti (2011) to study the effects of local demand shocks on local labor

⁷The revenues from the royalties accrue for 30 per cent to the national level (which devoted it entirely to the regional administration since 1999), 55 per cent to the regional level, and 15 per cent to municipalities. The revenues generated by each specific mining concessions are distributed across the municipalities as follows: 20 per cent accrues to the municipalities where the main collection and treatment center is located (Viggiano in our case), and the rest to municipalities where oil fields are located (five in our case), proportionally to the numbers of oil wells.

 $^{^{8}}$ In real terms the Brent price was equal to 28.6 2010 US dollars in 1990, to 31.9 in 2001 and to 91.4 in 2014, the last year in our sample.

markets (Greenstone et al., 2010; Aragón and Rud, 2013; Allcott and Keniston, 2018; Pelzl and Poelhekke, 2020). The same framework can be used to guide our analysis.

The main focus of our work is to study the employment effect of the establishment of an oil extraction facility, which can be interpreted as a local labor demand shock, and the heterogeneity of the effect in terms of firm characteristics. We study a capital intensive industry, so in principle the *direct* effect of the new plant on employment is expected to be small, because the oil company hires only few workers. The new establishment may increase demand for goods and services in the upstream sector, namely for those local firms selling goods and services to the oil company, thus increasing the demand for labor by the upstream sector (*first-order indirect* effect). This effect is likely sizable during the initial investment phase, that is the main focus of our analysis. To the extent that the *direct* and *first-order indirect* effects increased employment and that the average real wage does not decline (for example due to an increase in local prices such as housing), then the purchasing power of the local population increases, and thus spending as well. This would translate into a raise in local labor demand, as long as the propensity to consume locally-produced good is high enough. This second-order indirect effect would thus reinforce the first-order indirect one. Notably, the positive labor demand shock may well involve responses both at the extensive margin (employment) and at intensive margin (hours worked). The effect at the intensive margin is expected to be particularly relevant in the short run, because hours can be adjusted more quickly than employment (Rosen, 1968; Becker, 1962).

In this work, we also study the effect on wages. According to the model by Moretti (2011), a labor demand shock will increase salaries, because it reduces the monopsony power of firms (Manning, 2003): the upward pressure on wages is higher the more the labor supply is inelastic across areas, and the less workers are substitutable across sectors (due to migration from other locations and reallocation from other sectors). Those predictions are derived from full-employment models, that effectively assume away involuntary unemployment. However, the Italian context is characterized by high unemployment and high wage rigidity, with unions negotiating sector-specific

wages that are very similar across regions and firms. Wages in national contracts are typically set close to the equilibrium level in Northern regions, where firms are more productive than in the South (Bodo and Sestito, 1994). Decentralized firm-level bargaining is allowed, but very rare in the South and more frequent in large firms in the North, which contributes to create a small wage gap across areas (Casadio, 2010). These features imply that in the South increases in labor demand do not translate into higher wages (which are not flexible, and are already above the market-clearing level (Accetturo et al., 2022)), but rather into higher employment, drawing from the reserve of unemployed workers (Boeri et al., 2021). Empirically, this results in a low sensitivity of wages to shocks in Italian labor markets, especially in the South (Ciani et al., 2019).

5 Data

Social security data The analysis at the firm level is based on administrative microdata from the National Social Insurance Institute (INPS), covering all private sector firms (except agriculture) with at least one employee in a month of the year.⁹ The frequency is at annual level and the sample covers the period from 1990 to 2014. In the data, the observational unit corresponds to a social insurance position at INPS. In each year, a single social position (observation) is associated to a firm, but some firms have multiple social positions per year. This is due to administrative reasons, related for instance to the distinction of employees with different national contracts or to the presence of multiple local autonomous administrative units of the same firm. For each social insurance position, we have information on the number of employees for different job positions (blue collars, white collars, apprentices, managers, not classified), together with information on the total payroll. Dividing this by the number of employees we obtain a measure of gross wage per worker. Given that we observe average annual employment, the smallest social insurance position has employment equal to 1/12 (i.e. one employee in a single month of a given year). The data in-

⁹For an example of recent research using this data see Citino and Linarello (2022).

clude information on the industry (available as 2-digits ATECO code, corresponding to 2-digits NACE) and location at the municipal level.

First, we select social insurance positions located in four provinces (NUTS 3): Potenza (where the oil fields are) and the three neighboring provinces of Matera, Salerno and Cosenza.¹⁰ Second, we aggregate variables at the firm level assuming that the location and the industry of each firm coincides with the observation with highest employment; in our sample, the share of multi-position firms is 4 per cent. Finally, we drop firms that change municipality within our time period (approximately 3 per cent). The final sample is thus an unbalanced panel of firms, because of entry and exit in the market and/or in the sample area. A final note is warranted: the oil plant under investigation belongs to ENI, a large oil company with headquarter outside our sample region; as such, our data does not consider the employees working for the oil company. Thus, all our estimates do not include the *direct* effect, which corresponds to approximately 300 employees (ENI, 2012).

Distances from the oil centre We use data from the National Statistical Institute (Istat) on travel distances by car between municipalities, measured in kilometers or hours (Matrici di pendolarismo). Distances are measured from the municipality of Viggiano, which is the one hosting the oil treatment center. In our sample, the two measures of distance (kilometers and hours) the oil center are very highly correlated (0.96) and such that it takes on average one hour to drive 100 km (60 km in the direct vicinity of the oil fields). For sake of brevity, we only present results using time distances, but they are almost identical to the one obtained using the other measure. Given that distances by car can be affected by the treatment if the latter leads to road construction, we also use for robustness measures of air distances.

Tradable and upstream sectors When studying the heterogeneity of the effect of the oil shock, we classify firms according to their industry. We split firms in tradable and untradable according to the standard and broad definition used in AMECO

 $^{^{10}{\}rm Potenza}$ and Matera are the only two NUTS3 provinces of Basilicata, while Salerno is a province in Campania and Cosenza is in Calabria.

(2021). The AMECO database classifies the ATECO (NACE) sectors A to E and G to I (agriculture and fishing, mining and utilities, manufacturing, trade, hotels, communications) as tradable, while sectors F, J to P (construction, finance and business services, market services, other service activities) are considered as non-tradable. We also classify firms according to their degree of upstreamness with respect to the oil industry, following the approach in Allcott and Keniston (2018). For a given industry *i*, upstreamness is measured as the share of value added purchased by the oil industry from *i* out of total value added produced by industry *i*. The proxy is calculated from national input-output tables by Istat and it varies at the ATECO 2-digit level. According to this definition, the most upstream industries include water supply, rental and leasing activities, manufacture of chemicals, manufacture of non metallic minerals, waste management, oil production itself.¹¹

Census data For the aggregate analysis at the municipal and local labor market level, we turn on data from the decennial population censuses by Istat. This source has comprehensive information at the municipal level on resident population and employment by gender and sector, and it is available for years 1951, 1961, 1971, 1981, 1991, 2001 and 2011.

6 Firm level analysis

Our empirical design for the firm level analysis is very close to what Aragón and Rud (2013) use to study the effects of a single large gold mine in Peru. We estimate by OLS several difference-in-differences specifications akin to model (1), where i is the index for firm, m for municipality, and t for year.

$$Y_{imt} = \alpha + \beta D_m \times Post_t + \gamma D_m + \delta Post_t + \theta X_{imt} + \varepsilon_{it}.$$
 (1)

¹¹Table 18 reports the degree of upstreamness to the oil production of each industry.

The vector X_{imt} includes control variables. The coefficient of interest β is attached to the interaction between a cross-sectional variable D_m that measures the vicinity to the oil fields (i.e. to the municipality of Viggiano), and a dummy variable $Post_t$ for the development of the extractive industry. In particular, $Post_t$ takes the value zero between 1990 and 1995, and one between 1996 and 2001. In lack of complete annual data on investment by the oil company, our binary definition allows to capture the effect of the initial material investment made to set up the oil industry in Val d'Agri. In fact, the years 1990-95 can be considered as an exploratory period, with little production and investment. The following six years mark instead a period of significant material investment: the construction of the pilot treatment center in 1996 and its expansion until reaching a capacity of more than 100,000 barrels per day in 2001 (1.5 billions euro; see section 3 for more details). Our baseline analysis is restricted to the period 1990-2001, a time in which production and the related royalties were low, so our specification identifies only the effect of the initial investment (and the expectations of the forthcoming windfall of resources). In subsequent analyses, we extend the sample to 2013, to investigate the persistence of the effects beyond 2001.

We experiment with several definitions of the variable D_m that measures vicinity to the oil fields. At first, we use a binary definition that considers as treated all firms located in municipalities in the same local labor market (LLM) as Viggiano, where the oil treatment center is located.¹²

Firms located in the treated area are somewhat different from those located outside, as from the summary statistics reported in Table 17 for the period before the construction of the first oil treatment center in Viggiano (1990-1995). Most notably firms are on average smaller, have lower wages and the sectoral composition is slightly different. However, the difference in firms size is due to few large firms that are only present in the control group: after trimming the right tail, this difference shrinks.

Looking at the histograms of the number of employees and of the average wage, the distributions are remarkably similar between the treated and control groups. Figure 4

¹²The municipalities are: Grumento Nova Marsicovetere, Marsico Nuovo, Moliterno, Montemurro, Paterno, San Chirico Raparo, San Martino d'Agri, Sarconi, Spinoso, Tramutola e Viggiano.





Note: black borders are municipal boundaries; grey municipalities are those included in the sample; orange municipalities are those in the local labor market where the oil treatment center is located.

| | Ctrl | Treat | Diff |
|---------------------------------|-----------------------|--------|---------------|
| Average wage (euro) | 122822 | 120076 | 27.45** |
| Average wage (euro) - trim 99.9 | 122360 | 120076 | 22.84^{**} |
| Average wage (euro) - trim 99 | 120568 | 119318 | 12.50 |
| Employees | 5.06 | 2.99 | 2.08^{*} |
| Employees - right trim 99.9 | 4.51 | 2.99 | 1.53^{***} |
| Employees - right trim 99 | 3.66 | 2.79 | 0.87^{***} |
| Blue collars | 3.43 | 2.42 | 1.01^{**} |
| White collars | 1.05 | 0.37 | 0.67^{*} |
| Apprentices | 0.22 | 0.15 | 0.07^{***} |
| Manufacturing $(0/1)$ | 0.22 | 0.19 | 0.03^{***} |
| Construction $(0/1)$ | 0.30 | 0.44 | -0.14^{***} |
| Trade $(0/1)$ | 0.24 | 0.20 | 0.04^{***} |
| Tourism $(0/1)$ | 0.06 | 0.05 | 0.01^{**} |
| Other services $(0/1)$ | 0.17 | 0.12 | 0.05^{***} |
| Tradeble $(0/1)$ | 0.57 | 0.49 | 0.08^{***} |
| Upstream $(0/1)$ | 0.23 | 0.22 | 0.01 |
| No. observations | 181592 | 3191 | |

Table 2: Descriptive statistics. Sample: 1990-1995.

plots the distribution of firm size (in terms of employment) in the local labor market where Viggiano is located (in black) and in the rest of the sample (in red) in the period 1990-1995. In the left panel, the variable is trimmed at the 99.9th percentile, while on the right panel at the 99th percentile to visualize the distribution better. The distributions in the treatment and control groups are quite similar: approximately 40 per cent of the observations have at most one employee, 20 per cent between one and two, 10 per cent between two and three, 5 per cent between three and four, and so on with a long right tail. The control group has a longer tail, with few firms having more than 6,000 employees; on the contrary, the largest firm in the treated group has approximately 600 employees. Figure 5 plots the same for the average wage, trimmed at the 0.1 per cent and 99.9 per cent percentile on the left, and on the 1 per cent and 99 per cent percentiles on the right panel.

Figure 4: Firm distribution: number of employees



Note: treated refers to firms in the same local labor market (LLM) as Viggiano, and controls to those outside. In the left panel, the number of employees variable has been right trimmed at the 99.9th percentile; in the right panel, at the 99th percentile. The width of each bar is 1 employee.

As an alternative binary definition, we also consider as treated those municipalities within 30 minutes by car from Viggiano, but the set of included municipalities is very similar.¹³

Next, we use travel distance from Viggiano to the municipality where the firm is located as a continuous measure of treatment. As distance enters linearly in our specification, this definition assumes that the treatment effect decays linearly as firm locations move away from Viggiano.

¹³This definition also includes Missanello, Laurenzana and Armento but excludes San Chirico Raparo.



Figure 5: Firm distribution: average wage

Note: treated refers to firms in the same local labor market (SLL) as Viggiano, and controls to those outside. In the left panel, the average wage has been trimmed at the 99.9th percentile; in the right panel, at the 99th percentile.

Finally, we use a spline of travel distance. We divide firms in groups according to their distance from Viggiano. The first group includes firms located in municipalities within 30 minutes from Viggiano. We then group remaining firms in brackets of further 30 minute distances, with the last category containing all firms located between 120 and 166 minutes from Viggiano. We also experiment with finer categories: the first group include only Viggiano, the second municipalities within 16 minutes, and then we categorize remaining firms in brackets of further 16 minute distances, with the excluded category containing all firms located more than 64 minutes away.¹⁴ These exercises are useful in order to provide evidence on the presence of geographical spillovers, namely on the possibility that an increase in the outcome variable for firms close to the plant is mirrored by a corresponding, or even larger, drop for those located further away, for example due to the reallocation of workers.

6.1 Identification

The coefficient β in (1) identifies the average treatment effect on the treated under the standard parallel trend assumption: in a counterfactual scenario without the birth and expansion of the oil industry, firms located close to Viggiano would have displayed the same dynamics in employment. The assumption is untestable, but it is possible to provide suggestive evidence in support of its validity, by showing that the outcome

¹⁴Defining categories over a 15-minute interval would imply the second category to include only very few firms in a single municipality.

variables display similar trends in the two groups in the pre-treatment period. If the outcome variables are also similar in terms of levels and distribution, this would further corroborate the validity of the identifying assumption.

Notably, in the DID setting the choice of the functional form of the outcome variables (e.g. employees in logs or in level) is crucial, because the parallel trend assumption can not hold simultaneously for all functional forms, except in some special cases (Angrist and Pischke, 2008).¹⁵ In our case, the distributions of employment and wage are rather similar in the treated local labor market and elsewhere, but the control group has a much longer right tail, as discussed above. Thus it is not ex-ante clear if the parallel trend holds (or not) independently on the functional form of the outcome. We thus experiment with both outcomes in levels and logs.

To test for parallel trends in the pre-treatment period, we estimate the event-study model:

$$Y_{ismt} = \sum_{t=1990}^{2008} \beta_t D_m \times FE_t + FE_m + FE_s + FE_t + \varepsilon_{ismt}, \qquad (2)$$

where *i* is the index for firm, *m* for municipality, *s* for sector and *t* for year. Here we code D_m as a binary variable corresponding to the local labor market where the oil fields are located. The excluded year is 1995. Starting with the number of employees as outcome variable, Figure 6 plots the coefficients β_t : on the left with the outcome in levels, and on the right with the outcome in logs. Figure 7 is the corresponding plot for the average wage.

For both outcome variables in levels, the event-study specification suggests a parallel trend in the pre-treatment period (before 1996), and positive effects afterwards. Thus in our baseline estimates we use the outcome in levels, a case for which we are able to make causal statements on solid grounds (this is the approach suggested by Cunningham (2021)). In the case of the logged outcomes, the coefficients display a somewhat different pre-treatment trend: firms in the treated LLM are on a slightly

¹⁵The classical view is that the parallel trend assumption can hold for different functional forms if the distribution of the outcome variable in the pre-treatment period is the same in the treatment and in the control group. However, this view has been recently challenged by Rothy and Sant'Anna (2021), who show that having identical baseline distributions is not necessary (nor sufficient) for the parallel trends assumption to be invariant to transformations.



Figure 6: Event study specification, firm-level. Effects on the number of employees.

Note: the plots report the coefficients on the interaction between year dummies and a dummy for the local labor market where oil extraction takes place. The excluded year is 1995. The model also includes year dummies, sector dummies and municipal dummies. The outcome variable is the number of employees in the left panel, and its log in the right panel.

downward trajectory relative to the control group; we believe that this pattern is not worrisome because it would result in an underestimation of any positive effects, but we anyway relegate the log-model to the robustness section.



Figure 7: Event study specification, firm-level. Effects on the average wage.

Note: the plots report the coefficients on the interaction between year dummies and a dummy for the local labor market where oil extraction takes place. The excluded year is 1995. The model also includes year dummies, sector dummies and municipal dummies. The outcome variable is the average wage in the left panel, and its log in the right panel.

6.2 Estimated effects on the firm-level number of employees

We start by estimating equation (1) by OLS using firm-level data over the period 1990-2001, and using a dummy D_m equal to one for firms in the LLM where the oil

fields are located. Results are reported in table 3. Column (1) shows the results for the specification without any control variables; in column (2) we add municipality fixed effects. In the following columns, we add different sets of dummies to account for preexisting differences in the distribution of firms by sector and to account for differential trends across sectors. In particular, the model in column (3) includes, together with municipality fixed effects, also (broadly defined) sector fixed effects (manufacturing, construction, trade, tourism, other services). Column (4) includes, as an alternative, more disaggregated industry fixed effects (two digits economic activity ATECO code); finally column (5) includes, in addition, the interactions of these fixed effects with the *Post* dummy. Standard errors are clustered at the LLM level (56 groups).¹⁶

| | Number of employees | | | | | |
|------------------------|------------------------|------------------------|------------------------|------------------------|------------------------|--|
| | (1) | (2) | (3) | (4) | (5) | |
| Post * Treat | 0.86^{***} (0.09) | 0.97^{***} (0.08) | 0.97^{***} (0.09) | 0.91^{***} (0.08) | 0.83^{***} (0.07) | |
| Baseline outcome | 3.0 | 3.0 | 3.0 | 3.0 | 3.0 | |
| Relative effect | 0.29 | 0.32 | 0.33 | 0.31 | 0.28 | |
| FE municipality | | Х | Х | Х | Х | |
| FE sector | | | Х | | | |
| FE NACE-2-digit | | | | Х | | |
| FE NACE-2-digit X Post | | | | | Х | |
| N firms | 82020 | 82020 | 82020 | 82020 | 82020 | |
| N observations | 393646 | 393646 | 393646 | 393646 | 393646 | |

Table 3: Effect of oil industry on firm size (number of employees)

Note: the table reports the coefficient of the interaction between a dummy equal to one for the years 1996-2001, and a dummy equal to one for the firms in the local labor market where the oil extraction takes place. The specification always include also the two terms of the interaction separately, so it amounts to a difference-in-differences specification, and thus the coefficient of the interaction corresponds to the average treatment effect on the treated. The outcome variable is the number of employees (right trimmed at the 99.9th percentile), whose average in 1990-95 period for the treatment group is reported as baseline outcome. The table also reports the relative effects that is the ratio between the coefficient and the baseline outcome. Different columns correspond to the inclusion of different sets of fixed effects. The sample covers the years 1990-2001. Standard errors in parenthesis are clustered at the local labor market level.

The coefficient of interest, representing the average effect of the treatment on the

¹⁶We also experiment with standard errors that account for arbitrary spatial correlation with a spatial cut-off of up to 200 kilometers (Colella et al., 2019). Standard errors are of similar magnitude, and become smaller as the spatial cut-off increases. Since calculating these standard errors is computationally very demanding, we always report standard errors clustered at the LLM level.

treated firms, is estimated in the range 0.83-0.97, depending on the exact model, and it is always statistically significant at the 99 per cent level. In our preferred specification (column 3), the effect equals 0.97.¹⁷ Hence, our estimates imply that, on average, treated firms increase employment by roughly one unit as a consequence of the treatment. The effect is quite large, considering that average employment in the 1990-95 period for the treated firms (what we call "baseline employment" in the tables) is 3 units: it means that treated firms increase their employment by 33 per cent (reported as "relative effect" in the table).¹⁸

Next, we turn to results obtained with treatment status defined using travel distance by car from the municipality that hosts the oil treatment center. When firms are defined as treated if they are within 30 minutes from Viggiano, the coefficient of interest is very similar, around 0.9; this is not surprising, because this definition approximately corresponds to the local labor market, as noted above.¹⁹ When distance enters linearly as a continuous variable, the coefficient is very small and not significantly different from zero.²⁰ This null result is due to the fact that the effect of the treatment is highly non-linear over space and very concentrated close to the oil fields. This emerges quite clearly when we group firms in bins of travel distance from the oil treatment center. Figure 8 provides a graphical illustration of these estimates. In the left panel, we consider a set of 30-minutes bins of travel time, and the excluded category is made by firms located more than two hours away from Viggiano. We only

¹⁷Estimates in columns (3) and (4) are very similar. In column (5) including the interactions of ATECO fixed effect with the treatment dummy implies that the effect of the treatment does not capture changes of employment due to the reallocation of firms across ATECO codes, which is an important component of the effect of interest.

¹⁸We do not report results from models with firms fixed effects. The OLS and fixed effects estimators of the DID parameter are numerically identical if the panel is balanced; if the panel is unbalanced, as in our case, the fixed effect estimator exploits only variation from the firms that are observed in both periods and hence produce an estimate which does not consider firms entering or exiting the sample (Lechner et al., 2016). As we are also interested in these firms, we prefer the model without firm fixed effects. For completeness, the DID coefficient estimated including firm fixed effects in model (1) is 0.43 and is statistically significant at the 99 per cent level. We interpret this result as suggesting that the increase of employment is due partly to already existing firms and partly to net natality of firms.

¹⁹Also the event-study graph looks very similar; see appendix A.

 $^{^{20}}$ The point estimate implies that an additional hour of distance from Viggiano *increases* employment by 0.03 units.

find a positive effect for firms in the first bin, while it is exactly zero at higher distances. This pattern suggests the absence of negative spillovers beyond 30 minutes, namely that the increase in firm size estimated in Val D'Agri is not mirrored by a corresponding drop in the surrounding areas. In the right panel, we consider a finer set of distance bins (16 minutes), and the excluded category is made by firms located more than 64 minutes away. It is possible to see that the effect is not only localized in the municipality where the oil center is located (first bin), but it decays linearly until it becomes zero beyond 32 minutes. Event-study graphs, reported in appendix A, are consistent with the presence of parallel trends in the pre-treatment period in all 30-minutes (Figure 12) and 16-minutes bins (Figure 13).

Figure 8: Effect on firm size (number of employees) at different distances



Note: the table reports graphically as dots the coefficient of the interaction between a dummy equal to one for the years 1996-2001, and variables capturing the presence of the oil industry: in the left panel a set of dummies for firms located in different 30-min bins of distance from Viggiano, where the excluded category is made by firms located more than two hours away; in the right panel a set of dummies for firms located in different 16-min bins of distance from Viggiano, where the excluded category is made by firms located more than 64 minutes away. The specifications always include municipality and sector fixed effects, so they amount to difference-in-differences. The outcome variable is the number of employees (right trimmed at the 99.9th percentile). The sample covers the years 1990-2001. The bars correspond to 95 per cent confidence intervals constructed using standard errors clustered at the LLM level.

Heterogeneity In this section we assess if the average treatment effect estimated above is constant across firms which are observationally different, focusing in particular on two attributes: firm size and sector. Moreover, we explore how the effect varies for different categories of employees. For each heterogeneity test presented in this section, we report in Appendix A the corresponding event-study graphs aimed at assessing the presence of parallel trends before 1996. We start by estimating the baseline model (1) on different sub samples made of firms of different size. The upper panel of table 4 shows that the treatment effect is positive and significant for all size groups, and it is increasing in size: smaller firms (0-5 employees) increase the number of employees by 0.04, while the effect for the largest firms (at least 50 employees) is 58.9. Scaling the effect using the pre-treatment employment yields a relative effect of 1 or 2 per cent for firms below 50 employees and of 57 per cent for the larger firms.²¹ As a second exercise, we estimate the baseline

| | N of empl | oyees. Subs | amples: em | ployees in: |
|--|---|---|--|--|
| | $(1) \\ 0-5$ | (2) 5-10 | (3) 10-50 | $(4) \\ 50+$ |
| Post X LLM Oil | 0.032^{***} (0.008) | $\begin{array}{c} 0.073^{***} \\ (0.019) \end{array}$ | 0.276^{**} (0.119) | 57.696^{***} (2.966) |
| Baseline outcome Relative effect N firms N observations | $1.54 \\ 0.02 \\ 78180 \\ 323305$ | $6.94 \\ 0.01 \\ 12945 \\ 38278$ | $17.14 \\ 0.02 \\ 6933 \\ 28839$ | $100.38 \\ 0.57 \\ 809 \\ 3224$ |
| | Pr(nu | mber of emp | oloyees in ir | nterval) |
| | | | | |
| | (1) 0-5 | (2) 5-10 | (3) 10-50 | $(4) \\ 50+$ |
| Post X LLM Oil | $(1) \\ 0.5 \\ -0.024^{***} \\ (0.002)$ | $(2) \\ 5-10 \\ 0.012^{***} \\ (0.001)$ | $(3) \\ 10-50 \\ 0.007^{***} \\ (0.002)$ | $(4) \\ 50+ \\ 0.005^{***} \\ (0.000)$ |

Table 4: Heterogeneity by size

Note: the table reports the coefficient of the interaction between a dummy equal to one for the years 1996-2001 (Post), and a dummy equal to one for the firms in the local labor market where the oil extraction takes place. In the upper panel, each column refers to a regression on the subsample of firms of the size reported above each column, and the outcome variable is the number of employees. In the lower panel, the sample is always full and the outcome variable is a dummy equal to one if a firm belongs to size category reported above each column. Both panels report the average of the outcome variable in the 1990-95 period for the treated observations, and the relative effects equal to the ratio between the coefficient and the baseline outcome. The sample always covers the period 1990-2001, and municipal and sector dummies are always included. Standard errors in parenthesis are clustered at the local labor market level.

 $^{^{21}}$ The event-study specification does not support a causal interpretation for the group of largest firms (see Figure 15 in Appendix A).

model (1) four times on the full sample, using as outcome variable a dummy equal to one if the firm belongs to each one of the four size categories. The results, reported in the lower panel of table 4, show that the the treatment has reduced the probability of a firm being small (0-5 employees) by 2.4 percentage points (from a baseline of 87 per cent). On the contrary, the treatment effect is positive for all other groups: the share of firms between 10 and 50 employees has increased by 0.7 percentage points (from 5 per cent), and the share of larger firms (50 employees or more) has almost doubled as a consequence of the treatment (from 0.3 per cent). Results from the eventstudy specifications are rather noisy but, especially for the 0-4 and 50+ categories, are broadly consistent with a causal interpretation of the results (see Figure 16 in Appendix A).

We have conducted a similar analysis to study heterogeneity across sectors. The effect of the treatment is positive in all sector sub-samples, as shown in the upper panel of table 5. However, as for size categories, some differences emerge. The effect is particularly strong in manufacturing (50 per cent, column 1). A positive effect in manufacturing is in line with previous evidence (Allcott and Keniston, 2018; Pelzl and Poelhekke, 2020). The second most affected industry is the tertiary sector, where the effect is approximately 30 per cent (columns 3, 4 and 5). While some of these effects might be due to demand by the oil company, the magnitude and pervasiveness of the increase suggests that second-order indirect effects are at least partly at work. Interestingly, the effect is positive also for touristic activities (column 4), where one might expect that environmental externalities would scare off tourists; however, the development of the new industry has likely generated a profitable local market for business travels, which fulfills the demand from those workers who stay on site on a temporary basis. Finally, we also find a positive effect in the construction sector (12 percent, column 2); this is not completely surprising because the period under consideration is characterized by significant investment in physical capital, which also entails some buildings.²² Notice that across the columns of Table 5, the effect is

 $^{^{22}}$ Event-study specifications support a causal interpretation in all sectors (see Figure 18 in Appendix A)

stronger for the so-called tradable sectors: manufacturing and trade; this result is confirmed when we create a dummy for tradable sector according to the AMECO (2021) definition (manufacturing, trade and some specific other services) and interact it with $Post \times LLMoil$. At first, this finding might appear puzzling as the local labor market framework predicts smaller effects for the tradable sectors, because they are subject to competition from firms in other regions (Moretti, 2011); however, our result is consistent with the previous empirical literature, that has pointed that this crude binary partition does not capture whether a firm produces locally tradable goods or not, that is what matters in spatial models. For example, Allcott and Keniston (2018) show that even within manufacturing, some sub-sectors are much more tradable than others, and that this matters for consequences of local oil shocks. Unfortunately, we do not have the granular data that would allow us to replicate this type of analysis.

Next, we turn to the specifications where we use dummies for different sectors as left-hand side variables in equation (1). This approach estimates the change in the probability of belonging to a given sector, thus it can be thought as a test of sectoral reallocation. We do not find any changes in the share of manufacturing and trade firms (columns 1 and 3). On the contrary, the share of construction businesses declines by 3.4 percentage points, and there is a corresponding increase in the tertiary sector (1.6 percentage points in tourism, and 1.9 in other services).²³ These results suggest that if the "Dutch disease" is at play, it does not crowd out manufacturing, but rather the construction sector.²⁴

The theoretical framework predicts that the rise of the oil industry should affect first and foremost those firms that sell goods and services to the extractive sector. To test this prediction, we construct a continuous proxy for how much each firm is upstream to the oil industry using national input-output tables (see section 5). We

²³Note that the results from the event-study specifications are very noisy and do not always support a causal interpretation (see Figure 18 in Appendix A).

²⁴This finding might appear puzzling given that construction is usually considered a non-traded sector; however, at this geographical level, it is not obvious that for construction works it is necessary to rely on firms located in the same commuting zone, rather than in other surrounding labor markets. Manufacturing is usually considered as a tradable sector, but recent research has shown that within manufacturing there is considerable industry heterogeneity in tradability at the local level (Holmes and Stevens, 2014).

| | | Nur | nber of em | ployees | | | |
|--|-----------------------------------|---|---|---|---|--|--|
| | (1) Manuf. | (2) Constr. | (3) Trade | (4) Tour. | (5) Oth. serv. | | |
| Post X LLM Oil | $2.788^{***} \\ (0.235)$ | $\begin{array}{c} 0.327^{***} \\ (0.091) \end{array}$ | $\begin{array}{c} 0.446^{***} \\ (0.040) \end{array}$ | $\begin{array}{c} 0.480^{***} \\ (0.111) \end{array}$ | $\begin{array}{c} 0.747^{***} \\ (0.145) \end{array}$ | | |
| Baseline outcome Relative effect N firms N observations | $5.32 \\ 0.52 \\ 15902 \\ 84108$ | $2.80 \\ 0.12 \\ 23228 \\ 105909$ | $1.50 \\ 0.30 \\ 20337 \\ 99219$ | $3.27 \\ 0.15 \\ 6695 \\ 28761$ | $2.32 \\ 0.32 \\ 18272 \\ 75649$ | | |
| | Pr(firm in sector) | | | | | | |
| | | Pr | (firm in sec | ctor) | | | |
| | (1) Manuf. | (2) Constr. | (firm in sec (3) Trade | etor) (4) Tour. | (5) Oth. serv. | | |
| Post X LLM Oil | (1) Manuf. 0.000 (0.004) | (2) Constr. -0.034*** (0.006) | (firm in sec (3) Trade -0.001 (0.005) | (4) Tour. 0.016^{***} (0.001) | (5) Oth. serv. 0.019*** (0.003) | | |

Table 5: Heterogeneity by sector

Note: the table reports the coefficient of the interaction between a dummy equal to one for the years 1996-2001 (Post), and a dummy equal to one for the firms in the local labor market where the oil extraction takes place. In the upper panel, each column refers to a regression on the sub-sample of firms of the sector reported above each column, and the outcome variable is the number of employees. In the lower panel, the sample is always full and the outcome variable is a dummy equal to one if a firm belongs to the sector reported above each column. Both panels report the average of the outcome variable in the 1990-95 period for the treated observations, and the relative effects equal to the ratio between the coefficient and the baseline outcome. The sample always covers the period 1990-2001, and municipal and sector dummies are always included. Standard errors in parenthesis are clustered at the local labor market level.

split firms in four quartiles based on their upstreamness, and interact a set of dummies for these quartiles with $Post_t \times D_m$ in our baseline regression. We report the results graphically in figure 9, that shows how the time trends (i.e. difference in the number of employees between the two periods) for treated (blue) and untreated (red) firms vary at different quartiles of upstreamness. The vertical difference between the two lines is the difference-in-differences estimate at the corresponding quartile. The time trend is always negative for firms in the control group, independently on the degree of upstreamness of the firm. On the contrary, the time trend is approximately zero for treated firms whose goods and services are not heavily purchased by the oil industry



Figure 9: Effect of oil industry presence on firm size in the upstream sector

Note: the figure reports graphically the effects estimated in a specification which a triple interaction between a dummy equal to one for the years 1996-2001 (Post), a dummy equal to one for the firms in the local labor market where the oil extraction takes place, and a set of dummies for the quartiles of upstreamness. The blue line is the increase in firm size (number of employees) in the local labor market where the oil extraction takes place, while the red line refers to the rest of the sample. The vertical distance between the two lines is the difference-in-differences estimate for different levels of upstreamness. Dashed line are the 95 per cent confidence intervals. The specification also include the Post dummy, and sector, municipality and quartiles of upstreamness fixed effects.

(first quartile), and then it gradually increases with upstreamness, reaching the highest level for the fourth quartile. Thus the treated effect (that is the vertical distance) is increasing in upstreamness. The causal interpretation is supported by the event-study specifications (see Figure 18 in Appendix A). The finding that upstream firms are more affected by the establishment of the oil industry is confirmed by various robustness checks (not reported). In particular, we model upstreamness as a continuous variable, or as a dummy tagging firms whose upstreamness is above the median or in the highest quartile.

Finally, given that our data allows us to distinguish, to some extent, the employees according to their skill group, we estimate the baseline equation 1 using as dependent variable the employees in each skill group, instead of total employment. The total effect, reported in column (1) of table 6, is the sum of the effect over the skill groups. It emerges that firms have increased both blue (column 2) and white collars (column 3) as a consequence of the treatment (the parallel trend assumption seems to hold; see

Figure 18 in Appendix A). In absolute terms, the effect is larger for blue collars, but given that these represent a much larger share of the work-force, in relative terms the effect is larger for white collars. On the contrary, the effect is negative for apprentices, which might be consistent with treated firms switching to more stable work contracts for their employees. However, our data does not allow to test this conjecture.

| | | Number of employees | | | | | | | |
|--|---------------------------------|---------------------------------|-----------------------------------|------------------------------------|------------------------------------|--|--|--|--|
| | (1) Tot. | (2) Blue | (3) White | (4) Appr. | (5) Other | | | | |
| Post X LLM Oil | 0.97^{***} (0.09) | 0.53^{***} (0.18) | 0.09^{***} (0.03) | -0.03^{***} (0.01) | 0.38^{***} (0.10) | | | | |
| Baseline outcome Relative effect N firms N observations | 2.99 0.33 82020 393646 | 2.42 0.22 82027 393814 | $0.37 \\ 0.24 \\ 82027 \\ 393814$ | $0.15 \\ -0.18 \\ 82027 \\ 393814$ | $0.04 \\ 10.11 \\ 82027 \\ 393814$ | | | | |

Table 6: Effect of oil industry on employees of different types

Note: the table reports the coefficient of the interaction between a dummy equal to one for the years 1996-2001 (Post), and a dummy equal to one for the firms in the local labor market where the oil extraction takes place. In each model the outcome variable is the number of employees of the category reported above each column: total, blue collars, white collars, apprentices and others. The average of the outcome variable in the 1990-95 period for the treated group is reported as baseline outcome. The table also reports the relative effects that is the ratio between the coefficient and the baseline outcome. Standard errors in parenthesis are clustered at the local labor market level.

Robustness tests We assess the robustness of the results in several directions: we validate them with respect to an alternative control group of firms located in municipalities that are more similar to those in Val d'Agri; we present results from model where the outcome takes the logarithmic form; we estimate the effect of oil extraction using the changes-in-changes estimator (Athey and Imbens, 2006), whose identifying assumption does not depend on the functional form of the outcome variable.

The first robustness test is motivated by the pre-1996 differences in firm size, average wage, and other characteristics between the firms in Val d'Agri and those in the control group (see 17). Although a perfect balancing is not required for identification in the DID setting, a more similar control group would make the results less sensitive to the functional form of the outcome variable. So far, we compared treated firms to all those located outside the local labor market where the oil fields are (in the four surrounding provinces; see figure 3). The idea was to choose, as a control, firms that are, as far as possible, exposed to similar context factors and local shocks, with the exception of the development of the extractive industry. Here we consider instead a more restrictive control group, excluding firms that are located in municipalities that we consider very different from those in Val d'Agri. Table 7 shows the results: column (1) report the average effect from the baseline model for comparison; in column (2) we only consider firms located in municipalities with less than 6,000 residents (which is approximately the number of residents in the most populous municipality in the treated LLM); in column (3) we only include firms located in peripheral or semi-peripheral municipalities (according to the definition by Agenzia per la Coesione territoriale (2014)); in column (4) we only include firms located in mountainous municipalities (defined as the ones with altitude higher than 600 metres, which is the altitude of the lowest municipality in the oil LLM); in column (5) we exclude firms located in the LLM of Melfi, that hosts a very large car plant; finally in column (6) we consider only firms at the intersection of all these sub-samples. The last sub-sample includes roughly 5,400 firms, as opposed to 83,000 in the full sample. All sub-samples are more balanced in terms of the level of the outcome variables and of several covariates (as observed before 1996) compared to our main sample (see Appendix B). The effect of the treatment is still positive and statistically significant throughout the columns of table 7, but its size is sometimes smaller, and halves in column (6) compared to the baseline estimate; however, the relative effects is still economically sizable (18 per cent). Figure 14 in the Appendix shows that the assumption of parallel trend in the pre-treatment period is likely to hold also in these restricted samples, perhaps with the exception of the sample used for the regression in column (4).

As an additional robustness test, we estimate the baseline models using the outcome variable in logarithmic form. In this case the treatment induces a 7 per cent increase in the number of employees (standard error 1 per cent), approximately 1/4 of the relative effect estimated with the model in levels. There are two reasons why the log-model estimates a smaller effect than the level-model. First, as reported in Section 6.1, the

| | | Different samples | | | | | | |
|-------------------------------------|------------------------|---|--|--|---|--|--|--|
| | (1) Full | (2) Small towns | (3) Periphery | (4) Mountain | (5) No Melfi | (6) All above | | |
| Post X LLM Oil | 0.97^{***} (0.09) | 0.64^{***} (0.08) | 0.75^{***} (0.12) | 1.04^{***} (0.31) | 1.01^{***} (0.08) | 0.55^{***} (0.10) | | |
| Baseline outcome Relative effect | $3.0 \\ 0.33$ | $\begin{array}{c} 3.0\\ 0.22 \end{array}$ | $\begin{array}{c} 3.0\\ 0.25\end{array}$ | $\begin{array}{c} 3.0\\ 0.35\end{array}$ | $\begin{array}{c} 3.0\\ 0.34 \end{array}$ | $\begin{array}{c} 3.0\\ 0.18\end{array}$ | | |
| N firms N observations | $82020 \\ 393646$ | $23436 \\ 108109$ | $26383 \\ 124020$ | $16743 \\ 79778$ | $80019 \\ 384078$ | $5357 \\ 24374$ | | |

Table 7: Difference-in-differences estimates in sub-samples restricted by firm location

Note: the table reports the coefficient of the interaction between a dummy equal to one for the years 1996-2001, and a dummy equal to one for the firms in the local labor market where the oil extraction takes place. The specification always includes also the two terms of the interaction separately, so it amounts to a difference-indifferences specification, and thus the coefficient of the interaction corresponds to the average treatment effect on the treated. The outcome variable is the number of employees (right trimmed at the 99.9th percentile), whose average in 1990-95 period for the treatment group is reported as baseline outcome. The table also reports the relative effects that is the ratio between the coefficient and the baseline outcome. The sample covers the years 1990-2001. Different columns correspond to regression using subsamples with different geographical coverage: (1) full; (2) municipalities with less than 6,000 inhabitans; (3) municipalities categorized as peripheral area; (4) municipalities with altitude higher than 600 meters; (5) exclusion of the local labor market of Melfi; (6) all the previous restrictions together. Standard errors in parenthesis are clustered at the local labor market level.

log-models display a declining trend in the pre-treatment period (see Figure 7), and thus their estimates can be interpreted as a lower bound of the true effect. Second, the log-model implicitly gives less weight to the right tail of the outcome distribution; as discussed before, the distribution of the number of employees has a long right tail, and the effect is larger in large firms (see the upper panel of Table 4). Note that these two reasons are interconnected: as discussed above, the parallel trend assumption should hold independently of the functional form if the distribution of the outcome is similar between treatment and control groups. In our case the control group has a much longer right tail, so the parallel trend graphs are different if we use the outcome in level or in logs.

As a final robustness test, we estimate our treatment effect using the non-parametric estimator changes-in-changes (CIC) proposed by Athey and Imbens (2006). The CIC is a generalization of the standard DID and aims at identifying the treatment effects for each quantile of the outcome distribution, rather than only the average treatment effect.²⁵ For this reason, the parallel trend assumption must hold for each quantile of

²⁵Both models identify only the effects on the treated. The CIC works only for the standard two group - two period case, that is suitable for our application.

the outcome distribution, and the estimates do not depend on the functional form of the outcome variable. Figure 10 reports the treatment effects estimated with the CIC

Figure 10: Changes-in-changes model



Note: the plots shows as dots the average treatment effect (mean) and the treatment effects at different quantiles of the outcome variable, that is the number of employees. The bars are 95 per cent confidence intervals obtained using bootstrapped standard errors. The estimator is the changes-in-changes proposed by Athey and Imbens (2006) implemented using the Stata package cic by Kranker (2016).

for the 25th, 50th, 75th and 90th percentiles, and for the mean (that corresponds to the average effect). The results are consistent with our previous analyses: the effect increases with firm size, ranging from 0.08 employees for the 25th percentile (10 per cent in relative terms) to 1 employee for the 90th percentile (15 per cent in relative terms). The average effect is equal to 0.7 employees (23 per cent in relative terms), that is one third smaller than the average effect estimated with the standard DID, but still sizable.

6.3 Estimated effects on the firm-level average wage

As discussed in section 3, the development of the oil industry may affect positively both employment and wage, via an increase in labor demand. After having estimated the effect on employment in the previous section, here we estimate equation (1) using the average monthly wage in the firm as the outcome variable. An important caveat is that in our data we only observe the average wage per worker at the firm level, which is calculated as the ratio between total firm payroll and the number of workers. This measure corresponds to the product of the average hourly wage times the average number of hours worked in the firm. Hence we are not able to disentangle the impact on the hours worked from the impact on the hourly wage.

Table 8 shows the results: in our baseline specification (column 3) the effect is roughly 46 euros (in monthly terms) during the period 1996-2001; this amounts to a 4 per cent increase in relative terms. The effect becomes smaller in column (4) and also not statistically significant in column (5), where we add respectively ATECO 2-digit fixed effects, and their interaction with the *Post* dummy. The effect on the average wage is thus smaller than the effect on employment.

| | Average wage | | | | | |
|---|-------------------------|--|--------------------------|-------------------------|----------------------|--|
| | (1) | (2) | (3) | (4) | (5) | |
| Post X LLM Oil | $46.11^{***} \\ (8.03)$ | $\begin{array}{c} 44.07^{***} \\ (7.42) \end{array}$ | 45.43^{***} (7.56) | 27.55^{***} (7.83) | 7.69 (7.99) | |
| Baseline outcome Relative effect FE municipality FE sector | $1200.8 \\ 0.04$ | 1200.8 0.04 X | 1200.8 0.04 X X | 1200.8 0.02 X | 1200.8 0.01 X | |
| FE NACE-2-digit X Post N firms N observations | $81998 \\ 393440$ | $81998 \\ 393440$ | $81998 \\ 393440$ | л 81998 393440 | X 81998 393440 | |

Table 8: Effect on firm-level average wage

Note: the table reports the coefficient of the interaction between a dummy equal to one for the years 1996-2001, and a dummy equal to one for the firms in the local labor market where the oil extraction takes place. The specification always include also the two terms of the interaction separately, so it amounts to a difference-in-differences specification, and thus the coefficient of the interaction corresponds to the average treatment effect on the treated. The outcome variable is the average wage calculated as total payroll divided by the number of employees (trimmed at the 99.9th percentile), whose average in 1990-95 period for the treatment group is reported as baseline outcome. The table also reports the relative effects that is the ratio between the coefficient and the baseline outcome. The sample covers the years 1990-2001. Different columns correspond to regression using different combinations of fixed effects. Standard errors in parenthesis are clustered at the local labor market level.

As said, our outcome variable captures variation in both hours and hourly wage. However, previous studies have found low sensitivity of wages to local labor demand shocks in Italy, because of high wage rigidity and high structural unemployment (Boeri et al., 2021; Ciani et al., 2019). Hence, we suppose that the effect estimated in this section mostly captures the response in hours worked, but we cannot assess this directly.

An additional interesting note relates to the dynamic of the effect, captured by the event-study model in equation (2). Contrasting figure 6 and figure 7, we can see that the increase in employment is rather gradual, but quite persistent over time; on the contrary, the average wage increases immediately at the beginning of the investment period, and then slowly reverts back to the pre-treatment level. The differences in timing may reflect that in the short-run it is easier for firms to adjust hours, rather than hire new employees (Rosen, 1968).

6.4 The persistence of the effects

The findings from the event-study specification (equation (2)) suggests that the estimated effects on employment are rather persistent over time, while less so in the case of the average wage. In this section we test more formally for the persistence of the effects: we extend the sample until 2013, and interact the indicator for presence of the oil industry with three dummies for the periods 1996-2001, 2002-07, and 2008-13 (the excluded period is again 1995-2000; for the sake of symmetry each period is sixyear long). As mentioned before, the 1996-2001 period is marked by high investment but low production; in the following period 2002-07, production and royalties increase substantially, while material investment declines; finally the last period 2008-13 corresponds to a maturity stage, but crucially overlaps with the financial and sovereign crises. Table 9 shows findings for both the number of employees (first three columns), and the average wage (last three columns); columns (1), (3), (4) and (6) have sector fixed effects, columns (2) and (5) have a teco times post fixed effects; columns (3) and (6) restrict the sample to firms located in municipalities with similar characteristics (see section 6.2). The baseline results suggest that the effects on employment seem to persist for a long period, while the average wage returns to its pre-treatment level after six years. However, results from the restricted sample suggests that also the

| | Ν | Number of emp | loyees | | ge | |
|------------------------------|----------------------------------|----------------------------------|--------------------------------|----------------------------|---------------------------------|---------------------------------|
| | (1) Full sample | (2) Full sample | (3) Selected sample | (4) Full sample | (5) Full sample | (6) Selected sample |
| 1996-2001 X LLM Oil | 0.90^{***} | 0.77^{***} | 0.49^{***} | 49.00^{***} | 6.07 | -15.17 |
| 2002-2007 X LLM Oil | 0.63*** | 0.37*** | -0.09 | 29.27*** | -13.32 | -38.20* |
| 2008-2013 X LLM Oil | (0.10) 1.15^{***} (0.12) | (0.10) 0.88^{***} (0.12) | (0.14) 0.28^{*} (0.15) | (10.19) 6.58 (11.80) | (10.13) -37.46*** (11.47) | (20.97) -75.84*** (19.73) |
| FE municipality FE sector | XXX | X | XXX | XXX | X | XXX |
| FE NACE-2-digit X Post | | Х | | | Х | |
| N firms N observations | $156568 \\ 1012879$ | $156567 \\ 1012876$ | $9522 \\ 60114$ | $156529 \\ 1012570$ | $156528 \\ 1012567$ | $9521 \\ 60080$ |

Table 9: Persistence of the effect

Note: the table reports the coefficients of the interactions a dummy equal to one for the firms in the local labor market where the oil extraction takes place, and three dummies for the periods 1996-2001, 2002-2007 and 2008-2013 (the excluded category is 1990-1995). The specification always include also the terms of the interactions separately, so it amounts to a difference-in-differences specification. The sample period is 1990-2013. The outcome variable is the number of employees in the first three columns and the average wage in the last three columns. The average of the outcomes in 1990-95 period for the treatment group is reported as baseline outcome. The table also reports the relative effects that is the ratio between the coefficient and the baseline outcome. Different columns correspond to regression using different combinations of fixed effects. Standard errors in parenthesis are clustered at the local labor market level

effect on employment does not persist beyond 2001. All in all, we believe that these results must be interpreted with caution because, as more time passes from the initial establishment of the oil industry, more factors come into play, most notably the royalties windfall and the regional investment plan starting in the early 2000s (see section 3). Note that our reduced-form estimator can only capture the net effects of all these factors, and is not well suited for estimating effects far away in time.

7 Aggregate analysis

In the previous sections, we have focused on the micro effects and on the mechanisms, but we did not investigate much the consequences on the local labor market as a whole. However, estimating the effect on aggregate employment is interesting and important in its own sake, for example from the perspective of a policymaker.

The firm-level analysis suggests that the birth of the extractive industry increased firm size (in terms of number of employees) in all sectors; the average treatment effect is approximately 1 employee in our baseline firm-level specification (0.55 in our most conservative model). Since in our administrative data the treated LLM hosts approximately 400 firms in 1996, our estimates imply an increase in employment of roughly 400 units (220 in our most conservative model). To put this into perspective, this increase is approximately the same as the direct employment in the extractive facility. Since our data do not include workers employed by ENI (see section 5), our firm-level estimates suggest that each job in the oil company is coupled with approximately one additional job in other private firms.

A limitation of the back-of-the-envelope calculation above is that it assumes that the number of firms stays constant. This is not necessarily the case, as the number of active businesses can change as a consequence of the treatment or for other exogenous reasons. In this section, we address this issue by estimating directly the effect of our treatment on the number of firms (section 7.1).

A second and even more important feature of our firm-level social security data is that it only covers approximately 12 per cent of total employment in Val d'Agri. In fact, this dataset records approximately 1,200 firm employees, while the census records approximately 10,000 people with an occupation in the area. The difference is explained by the fact that the social security dataset does not include sizable categories such as self-employed workers, public employees, farmers, employees of firms with headquarter outside the sample (e.g. the oil company), and irregular workers. To estimate aggregate effects, we thus turn at data from the decennial population censuses (section 7.2).

7.1 Firm demographics

To estimate the effects of the establishment of the oil industry on the number of firms, we collapse the social security data at the level of LLM. We estimate an event-study specification similar to model (2); the unit of observation is the LLM, the outcome variable is the number of firms, and the coefficients of interest are those associated with the interactions between the dummy for the Val d'Agri labor market and the year dummies. We present the results graphically in Figure 11: the number of firms drops by approximately 100 units in the period 1996-2001, relative to the previous period and to the other local labor markets. The decline is sizable, but the figure clearly shows that the negative trend begins much earlier than 1996, and continues at



Figure 11: Event-study graph: number of firms

Note: the plot reports the coefficients on the interaction between year dummies and a dummy for the local labor market where oil extraction takes place. The excluded year is 1995. The model also includes year dummies and municipal dummies. The outcome variable is the number of firms. Confidence bars are based on robust standard errors.

a similar pace until 2008. As such, we do not consider this estimate as a causal effect induced by the oil industry. Nevertheless, this decline must be taken into account when quantifying the aggregate effects of the firm-level estimates, which must then be adjusted downwards.

7.2 Aggregate labor market effects

In order to estimate the aggregate labor market effects, we turn at data from the decennial population censuses that cover the entire economy. This source has comprehensive information at the municipal level on population and employment based on the residence of the workers. Census data covers all sectors, including self-employment, the public sector and agriculture, and even oil company workers who reside in the area. The data is available for all years ending with 1, starting in 1951. The timing is particularly convenient for our aim, as the 1991 edition collects information about the situation before the rise of the oil industry, and the 2001 edition covers the end of the period of massive investment by ENI.

First, it is useful to have a look at the economic structure in Val D'Agri from a descriptive point of view. Table 10 reports figures for the aggregate local labor

| Year | Pop. | Empl. | Employment rate | | | Emp | loymer | nt incie | lence |
|------|-------|-------|-----------------|------|------|------|--------|----------|-------|
| | | | Tot. | Mal. | Fem. | Agr. | Ind. | Tra. | Oth. |
| 1981 | 26957 | 10037 | 37.2 | 52.4 | 22.6 | | | | |
| 1991 | 28794 | 9758 | 33.9 | 47.9 | 20.6 | 19.3 | 33.0 | 31.9 | 15.8 |
| 2001 | 28509 | 10264 | 36.0 | 47.9 | 24.9 | 14.5 | 29.8 | 38.1 | 17.6 |
| 2011 | 27441 | 10604 | 38.6 | 50.1 | 28.0 | 12.5 | 27.5 | 42.3 | 17.7 |

Table 10: Dynamics in the oil local labor market

Source: different censuses, Istat. Note: the figures refer to the aggregate amount for the local labor market where the oil fields are located. Population refers to those aged 15 or older. Employment refers to all persons with an occupation. The employment rate is the ratio between the two. Employment incidence in different sectors is calculated as a share of employment.

market on population, employment, employment rate (also by gender), and incidence of different sectors. Between 1991 and 2001, active population²⁶ decreased by 280 people (-1 per cent), while the number of people with an occupation increased by 500 units (5 per cent); as a consequence, the employment rate increased from 33.9 to 36 per cent. Interestingly, the latter increase is driven by female, while the employment rate for male is constant in this time period. In terms of sectors, the employment shares in agriculture and manufacturing decline by, respectively, 5 and 3.3 percentage points; at the same time, the commercial sector grows from 31.9 to 38 per cent, and the remaining part of the service sector from 15.7 to 17.7 per cent. Of course these dynamics can not be attributed to the rise of the oil industry, at least not completely. For example, behind these numbers, we probably see an ongoing structural transformation away from agriculture, and the secular rise in female labor force participation (Del Boca, 2002). However, we can already see that the establishment of the extractive industry does not result in a large increase in aggregate employment, unless we think that in a counterfactual scenario without oil, employment would have declined dramatically.

In order to compare these numbers to a sensible counterfactual, we implement a difference-in-differences analysis at the municipal m level, similar to what already implemented at the firm level (equation 3). The main outcome variable is the employment rate²⁷, while the treatment dummy equals one for the 12 municipalities in

 $^{^{26}}$ Defined as aged 15 or more.

 $^{^{27}}$ Calculated with the population older than 15 years old at the denominator.

the Val D'Agri local labor market, and it is interacted with census year dummies. As before, we restrict our sample to municipalities in the four provinces Potenza, Matera, Salerno and Cosenza, making a total of 444 municipalities in 56 local labor markets (level at which we cluster the standard errors). We include years between 1971 and 2011, leaving 1991 as the omitted category. Importantly, we estimate our equation by weighted least squares, using population in 1991 as weight, in order to capture the employment effect in the local labor market as a whole. We fix population weights in the latest pre-treatment year in order to avoid endogenous weighting.

$$Y_{mt} = \sum_{t \in 1971, 1981, 2001, 2011} \beta_t ValAgri_m \times FE_t + FE_t + ValAgri_m + \varepsilon_{mt}, \qquad (3)$$

We are particularly interested in the coefficient on the interaction between the Val d'Agri dummy and the 2001 dummy, that identifies the change in the employment rate relative to the control group over the period of massive investment to set up the oil industry. Under a standard common trend assumption, this change can have a causal interpretation and be attributed to the raise of the extractive sector. The frequency of the census data does not allow to assess the pre-trend as convincingly as in the firm-level analysis, but we can still test whether the outcome variable was on a differential trend in 1981 and 1971. We also try to restrict the comparison to a control group of municipalities most similar to Val d'Agri; to do so, in one specification we also include the following controls interacted with year dummies: coastal dummy, mountain dummy, population class dummies, and a Basilicata dummy.

The results are reported in table 11: in the first two columns the outcome variable is the total employment rate, in the next two the male rate, and in the last two the female rate; even columns include controls. The analysis suggests that between 1991 and 2001 the total employment rate increased by approximately 1 percentage point, while it was not statistically different from 1991 in 1981 and in 1971. The effect in 2001 is statistically significant, but economically small, corresponding to approximately two hundreds more people.²⁸ When looking at the employment rate by gender, we see

 $^{^{28}}$ In particular the effect on the employment rate is driven for roughly two-thirds by an increase

| | Тс | Total | | ale | Fer | nale |
|--------------------|---------------------------|----------------------------------|-----------------------------|---------------------------------|----------------------------------|----------------------------------|
| | (1) | (2) | (3) | (4) | (5) | (6) |
| 1971 X LLM Oil | 0.23 (0.98) | -1.15 (0.82) | -2.14^{*} | -0.01 (1.26) | 1.74 (1.12) | -2.91^{**} (1.15) |
| 1981 X LLM Oil | (0.00) -0.15 (0.34) | (0.02) (0.26) (0.65) | (1.11) -1.15** (0.49) | (1.20) 0.88 (0.59) | 0.65 (0.45) | -0.53 (1.04) |
| 2001 X LLM Oil | (0.81) (0.82^{***}) | (0.00) 1.11^* (0.60) | (0.13) -0.31 (0.37) | (0.00) 0.27 (0.70) | (0.10) 1.98^{***} (0.28) | (1.01) 1.97^{***} (0.54) |
| 2011 X LLM Oil | (0.24) 0.32 (0.34) | (0.00) 1.61^{***} (0.53) | (0.57) 0.52 (0.50) | (0.76) 1.76^{**} (0.76) | (0.23) 0.31 (0.43) | (0.54) 1.64^{***} (0.47) |
| Controls N obs. | No 2216 | Yes 2216 | No 2216 | Yes 2216 | No 2216 | Yes 2216 |

Table 11: Effect on the employment rate in Val d'Agri

Note: the table reports the coefficient of the interaction between a dummy equal to one for municipalities in the local labor market where the oil extraction takes place, and year dummies (the excluded year is 1991). The specification always include year dummies and municipalities fixed effects, so it amounts to a difference-in-differences specification. The outcome variable is the employment rate. Even columns include controls interacted with year dummies: coastal dummy, mountain dummy, population class dummies, and a Basilicata dummy. Standard errors in parenthesis are robust to heteroskedasticity.

that the increase is entirely driven by females (2 percentage point increase in the employment rate, strongly significant), while the rate among males does not change (the coefficient is small, not significant, and changes sign when controls are added). These effects seem to persist, at least in part, until 2011, but in this respect the evidence is more mixed and depends on the control set, so we caution against over-interpreting these results.

| | Agriculture | | Industry | | Trade | | Other services | |
|----------------------------------|---|--|--|---|---|---|---|---|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) |
| 2001 X LLM Oil 2011 X LLM Oil | $\begin{array}{c} -0.01 \\ (0.23) \\ -1.24^{***} \\ (0.35) \end{array}$ | $\begin{array}{c} 1.30^{***} \\ (0.25) \\ 0.93^{**} \\ (0.44) \end{array}$ | $\begin{array}{c} -0.72^{***} \\ (0.19) \\ 0.38^{*} \\ (0.19) \end{array}$ | $\begin{array}{c} -1.07^{**} \\ (0.52) \\ 0.13 \\ (0.46) \end{array}$ | $ \begin{array}{c} 1.36^{***} \\ (0.12) \\ 1.12^{***} \\ (0.14) \end{array} $ | $ \begin{array}{c} 1.08^{***} \\ (0.18) \\ 1.14^{***} \\ (0.22) \end{array} $ | $\begin{array}{c} 0.19^{*} \\ (0.10) \\ 0.06 \\ (0.16) \end{array}$ | $\begin{array}{c} -0.20 \\ (0.12) \\ -0.59^{***} \\ (0.13) \end{array}$ |
| Controls N obs. | No 1332 | Yes 1332 | No 1332 | Yes 1332 | No 1332 | Yes 1332 | No 1332 | Yes 1332 |

Table 12: Effects on the employment by sector in Val d'Agri

Note: the table reports the coefficient of the interaction between a dummy equal to one for municipalities in the local labor market where the oil extraction takes place, and year dummies (the excluded year is 1991). The specification always include year dummies and municipalities fixed effects, so it amounts to a difference-in-differences specification. The outcome variable is the employment incidence of any given sector. Even columns include controls interacted with year dummies: coastal dummy, mountain dummy, population class dummies, and a Basilicata dummy. Standard errors in parenthesis are robust to heteroskedasticity.

in the numerator (employment), and for the rest by a drop in the denominator (population). To see this we re-estimate the model by using as outcome the employment rate where the denominator is always fixed at 1991 (coefficient not reported). To have a sense of the contribution of different sectors to the increase in employment, we compute the ratios of workers in each sector over active population (older than 15 years old), and use these as outcome variables. The results, reported in Table 12, suggest that the oil labor market experienced a decrease in industrial employment (which include both manufacturing and construction), and a corresponding and longlasting increase in the commercial sector, relative to the control group. These findings are consistent with the dynamics by gender emerged in the previous table. A possible explanation for the drop in the industrial sector relates to the relevant weight of construction, that employs only 25 per cent less workers than manufacturing in Val d'Agri (in the social security data), and that shrinks in relative terms (see table 5). Interestingly, there is also a positive effects on agriculture, although in that period the sector contracted in absolute terms (see 10).

8 Conclusion

What are the causal effects of opening a new large capital-intensive extraction facility on the local labor market? We answer this question in the context of a small peripheral area in the South of Italy, using firm-level and census data. Firms located close to the oil fields grow bigger in terms of employment compared to those further away. On average, employment increases by slightly less than 1 unit per firm, which corresponds to a 30 per cent increase in our sample. The effect is stronger for upstream industries and for large firms both in absolute and relative terms. We also find evidence of positive effects on average wages per worker, amounting to a 4 per cent increase in relative terms. The firm-level results do not imply a strong increase in employment at the aggregate level because the number of firms in the area is small and declining. This finding is confirmed by the census data that covers all sectors and worker types: the effect on the employment rate is approximately one percentage point. Our evidence implies that the plant opening triggered a reallocation of labor towards firms that are larger, and economically and geographically close to the entrant firm.

References

- Accetturo, Antonio, Giuseppe Albanese, Rosario Maria Ballatore, Tiziano Ropele, and Paolo Sestito (2022), "I divari territoriali in italia tra crisi economiche, ripresa ed emergenza sanitaria." Bank of Italy Occasional Paper.
- Accetturo, Antonio and Guido de Blasio (2019), Morire di aiuti. I fallimenti delle politiche per il Sud (e come evitarli). IBL Libri.
- Agenzia per la Coesione territoriale (2014), "I numeri della coesione." URL https://www.agenziacoesione.gov.it/comunicazione/campagne/i-numeri-della-coesione/.
- Allcott, Hunt and Daniel Keniston (2018), "Dutch Disease or Agglomeration? The Local Economic Effects of Natural Resource Booms in Modern America." *The Review* of Economic Studies, 85, 695–731.
- AMECO (2021), "Annual macro-economic database of the european commission's directorate general for economic and financial affair." URL https://ec.europa.eu/info/business-economy-euro/indicatorsstatistics/economic-databases/macro-economic-database-ameco_en.
- Angrist, Joshua D. and Jörn-Steffen Pischke (2008), Mostly Harmless Econometrics: An Empiricist's Companion. Princeton University Press.
- Aragón, Fernando M., Punam Chuhan-Pole, and Bryan Christopher Land (2015), The Local Economic Impacts of Resource Abundance: What Have We Learned? The World Bank.
- Aragón, Fernando M and Juan Pablo Rud (2013), "Natural resources and local communities: Evidence from a peruvian gold mine." American Economic Journal: Economic Policy, 5, 1–25.
- Athey, Susan and Guido W. Imbens (2006), "Identification and inference in nonlinear difference-in-differences models." *Econometrica*, 74, 431–497.

- Bank of Italy (2001), "Note sull'andamento dell'economia della Basilicata nel 2000." Technical report, Bank of Italy.
- Bank of Italy (2009), "L'economia della Basilicata nel 2008." Technical report, Bank of Italy.
- Basso, Gaetano (2017), "Local labor markets adjustments to oil booms and busts."
- Becker, Gary S. (1962), "Investment in human capital: A theoretical analysis." Journal of Political Economy, 70, 9–49.
- Berlingieri, Giuseppe, Sara Calligaris, and Chiara Criscuolo (2018), "The productivitywage premium: Does size still matter in a service economy?" AEA Papers and Proceedings, 108, 328–33.
- Bernard, Andrew B., J. Bradford Jensen, Stephen J. Redding, and Peter K. Schott (2007), "Firms in international trade." *Journal of Economic Perspectives*, 21, 105– 130.
- Black, Dan, Terra McKinnish, and Seth Sanders (2005), "The economic impact of the coal boom and bust." *The Economic Journal*, 115, 449–476.
- Bodo, Giorgio and Paolo Sestito (1994), "Squilibri territoriali nel mercato del lavoro e inflazione." *Caratteri strutturali dell'inflazione italiana, Il Mulino: Bologna*.
- Boeri, Tito, Andrea Ichino, Enrico Moretti, and Johanna Posch (2021), "Wage Equalization and Regional Misallocation: Evidence from Italian and German Provinces." *Journal of the European Economic Association*. Jvab019.
- Brandolini, Andrea and Matteo Bugamelli (2009), "Rapporto sulle tendenze nel sistema produttivo italiano." *Questioni di economia e finanza*, 45.
- Bubbico, Davide (2016), L'economia del petrolio e il lavoro : l'estrazione di idrocarburi in Basilicata tra fabbisogno energetico nazionale e impatto sull'economia locale.
 Ediesse, Roma.

- Bubbico, Davide (2019), "Chi trae vantaggio dalla ricchezza della Basilicata?" il Mulino, 956–964.
- Bugamelli, Matteo, Francesca Lotti, Monica Amici, Emanuela Ciapanna, Fabrizio Colonna, Francesco D'Amuri, Silvia Giacomelli, Andrea Linarello, Francesco Manaresi, Giuliana Palumbo, et al. (2018), "Productivity growth in italy: a tale of a slow-motion change." Bank of Italy Occasional Paper.
- Calligaris, Sara, Massimo Del Gatto, Fadi Hassan, Gianmarco IP Ottaviano, Fabiano Schivardi, et al. (2016), "Italy's productivity conundrum. a study on resource misallocation in italy." Technical report, Directorate General Economic and Financial Affairs (DG ECFIN), European Commission.
- Carrington, William J. (1996), "The alaskan labor market during the pipeline era." Journal of Political Economy, 104, 186–218.
- Casadio, Piero (2010), "Contrattazione aziendale integrativa e differenziali salariali territoriali: informazioni dall'indagine sulle imprese della Banca d'Italia." *Politica economica*, 26, 241–292.
- Ciani, Emanuele, Francesco David, and Guido de Blasio (2019), "Local responses to labor demand shocks: A re-assessment of the case of Italy." *Regional Science and Urban Economics*, 75, 1–21.
- Citino, Luca and Andrea Linarello (2022), "The impact of Chinese import competition on Italian manufacturing." *Review of International Economics*.
- Colella, Fabrizio, Rafael Lalive, Seyhun Orcan Sakalli, and Mathias Thoenig (2019), "Inference with arbitrary clustering." Technical report.
- Criscuolo, Chiara, Ralf Martin, Henry G Overman, and John Van Reenen (2019), "Some causal effects of an industrial policy." *American Economic Review*, 109, 48– 85.
- Cunningham, Scott (2021), Causal Inference: The Mixtape. Yale University Press.

- Cust, James and Steven Poelhekke (2015), "The local economic impacts of natural resource extraction." Annual Review of Resource Economics, 7, 251–268.
- Del Boca, Daniela (2002), "Low fertility and labour force participation of Italian women."
- Duranton, Gilles and Anthony J Venables (2018), "Place-based policies for development." Technical report, National Bureau of Economic Research.
- ENI (2012), "Eni in Basilicata." Technical report, ENI.
- Giroud, Xavier and Holger M Mueller (2019), "Firms' internal networks and local economic shocks." *American Economic Review*, 109, 3617–49.
- Greenstone, Michael, Richard Hornbeck, and Enrico Moretti (2010), "Identifying agglomeration spillovers: Evidence from winners and losers of large plant openings." *Journal of Political Economy*, 118, 536–598.
- Greenstone, Michael and Enrico Moretti (2003), "Bidding for industrial plants: Does winning a'million dollar plant'increase welfare?"
- Holmes, Thomas J. and John J. Stevens (2014), "An alternative theory of the plant size distribution, with geography and intra- and international trade." *Journal of Political Economy*, 122, 369–421.
- Iacono, Roberto (2016), "The Basilicata Wealth Fund: Resource Policy and Longrun Economic Development in Southern Italy." Climate Change and Sustainable Development 230604, Fondazione Eni Enrico Mattei (FEEM).
- Kline, Patrick and Enrico Moretti (2014), "People, places, and public policy: Some simple welfare economics of local economic development programs." Annu. Rev. Econ., 6, 629–662.
- Kranker, Keith (2016), "Effects of medicaid disease management programs on medical expenditures: Evidence from a natural experiment in georgia." Journal of Health Economics, 46, 52–69.

- Lechner, Michael, Nuria Rodriguez-Planas, and Daniel Fernández Kranz (2016), "Difference-in-difference estimation by FE and OLS when there is panel nonresponse." *Journal of Applied Statistics*, 43, 2044–2052.
- Manning, Alan (2003), Monopsony in Motion: Imperfect Competition in Labor Markets. Princeton University Press.
- Marchand, Joseph (2012), "Local labor market impacts of energy boom-bust-boom in western canada." Journal of Urban Economics, 71, 165–174.
- Marchand, Joseph and Jeremy Weber (2018), "Local labor markets and natural resources: asynthesis of the literature." *Journal of Economic Surveys*, 32, 469–490.
- Melitz, Marc J (2003), "The impact of trade on intra-industry reallocations and aggregate industry productivity." *econometrica*, 71, 1695–1725.
- Melitz, Marc J and Gianmarco IP Ottaviano (2008), "Market size, trade, and productivity." The review of economic studies, 75, 295–316.
- Michaels, Guy (2010), "The long term consequences of resource-based specialisation." The Economic Journal, 121, 31–57.
- Monte, Ferdinando, Stephen J. Redding, and Esteban Rossi-Hansberg (2018), "Commuting, migration, and local employment elasticities." *American Economic Review*, 108, 3855–90.
- Moretti, Enrico (2010), "Local multipliers." American Economic Review, 100, 373–77.
- Moretti, Enrico (2011), "Local labor markets." In *Handbook of Labor Economics*, 1237–1313, Elsevier.
- Munasib, Abdul and Dan S. Rickman (2015), "Regional economic impacts of the shale gas and tight oil boom: A synthetic control analysis." *Regional Science and Urban Economics*, 50, 1–17.

- Neumark, David and Helen Simpson (2015), "Place-based policies." In *Handbook of regional and urban economics*, volume 5, 1197–1287, Elsevier.
- Pelzl, Paul and Steven Poelhekke (2020), "Good mine, bad mine: Natural resource heterogeneity and dutch disease in indonesia." *CEPR Discussion Paper No. DP15271*.
- Percoco, M. (2012), "Mining Activity and the Setting up of New Businesses in Basilicata." Rivista economica del Mezzogiorno, 881–898.
- Rosen, Sherwin (1968), "Short-run employment variation on class-i railroads in the u.s., 1947-1963." *Econometrica*, 36, 511–529.
- Rosenthal, Stuart S and William C Strange (2020), "How close is close? the spatial reach of agglomeration economies." *Journal of economic perspectives*, 34, 27–49.
- Rothy, Jonathan and Pedro H.C. Sant'Anna (2021), "Local labor markets adjustments to oil booms and busts."
- Siegloch, Sebastian, Nils Wehrhöfer, and Tobias Etzel (2021), "Direct, spillover and welfare effects of regional firm subsidies." ZEW-Centre for European Economic Research Discussion Paper.
- Van der Ploeg, Frederick (2011), "Natural resources: curse or blessing?" Journal of Economic literature, 49, 366–420.
- Van Der Ploeg, Frederick and Steven Poelhekke (2017), "The impact of natural resources: Survey of recent quantitative evidence." The Journal of Development Studies, 53, 205–216.

Appendix A Additional event-study graphs



Figure 12: Event-study graph for the 30-min bins

Note: the plots report the coefficients on the interaction between year dummies and a dummy for municipalities at different distances from Viggiano estimated in the same regression. The excluded year is 1995 and the excluded distance bin is made of municipalities further than 120 minutes from Viggiano. The model also includes year dummies, sector dummies and municipal dummies. The outcome variable is the number of employees.



Figure 13: Event-study graph for the 16-min bins

Note: the plots report the coefficients on the interaction between year dummies and a dummy for municipalities at different distances from Viggiano estimated in the same regression. The excluded year is 1995 and the excluded distance bin is made of municipalities further than 64 minutes from Viggiano. The model also includes year dummies, sector dummies and municipal dummies. The outcome variable is the number of employees.



Figure 14: Event-study graph for the restricted geographical sub-samples

Note: the plots report the coefficients on the interaction between year dummies and a dummy for the local labor market where oil extraction takes place. The excluded year is 1995. The model also includes year dummies, sector dummies and municipal dummies. The outcome variable is the number of employees. The samples are as follows: "Small town" is restricted to firms in municipalities with less than 6,000 inhabitants; "Periphery" in peripheral municipalities; "Mountain" in municipalities above 600 meters; "Without Melfi" excludes the commuting zone of Melfi.

Figure 15: Event-study graph for heterogeneity by size (n. of employees - "intensive margin")



Note: the plots report the coefficients on the interaction between year dummies and a dummy for the local labor market where oil extraction takes place estimated in four separate regressions. The excluded year is 1995. The model also includes year dummies, sector dummies and municipal dummies. The outcome variable is the number of employees.



Figure 16: Event-study graph for heterogeneity by size (Pr(n. of employees) - "extensive margin")

Note: the plots report the coefficients on the interaction between year dummies and a dummy for the local labor market where oil extraction takes place estimated in four separate regressions. The excluded year is 1995. The model also includes year dummies, sector dummies and municipal dummies. The outcome variable is the number of employees.

Figure 17: Event-study graph for heterogeneity by sector (n. of employees - "intensive margin")



Note: the plots report the coefficients on the interaction between year dummies and a dummy for the local labor market where oil extraction takes place estimated in four separate regressions. The excluded year is 1995. The model also includes year dummies and municipal dummies. The outcome variable is the number of employees.



Figure 18: Event-study graph for heterogeneity by sector (Pr(n. of employees) - "extensive margin")

Note: the plots report the coefficients on the interaction between year dummies and a dummy for the local labor market where oil extraction takes place estimated in four separate regressions. The excluded year is 1995. The model also includes year dummies and municipal dummies. The outcome variable is the number of employees.

Figure 19: Event-study graph for heterogeneity by type of employee



Note: the plots report the coefficients on the interaction between year dummies and a dummy for the local labor market where oil extraction takes place estimated in four separate regressions. The excluded year is 1995. The model also includes year dummies, sector dummies and municipal dummies. The outcome variable is the number of employees.



Figure 20: Event-study graph for heterogeneity by quartile of upstreamness

Note: the plots report the coefficients on the interaction between year dummies and a dummy for the local labor market where oil extraction takes place. The excluded year is 1995. The model also includes year dummies, sector dummies and municipal dummies. The outcome variable is the number of employees.

Appendix B Additional tables

| Table 13: Descriptive statistics. Sample: small towns, 1990-1 | 1995 |
|---|------|
|---|------|

| | Ctrl | Treat | Diff |
|---------------------------------|-----------------------|--------|----------------|
| Average wage (euro) | 114085 | 120076 | -59.92^{***} |
| Average wage (euro) - trim 99.9 | 113594 | 120076 | -64.82^{***} |
| Average wage (euro) - trim 99 | 111606 | 119318 | -77.12^{***} |
| Employees | 3.88 | 2.99 | 0.89 |
| Employees - right trim 99.9 | 3.40 | 2.99 | 0.41^{*} |
| Employees - right trim 99 | 3.03 | 2.79 | 0.24^{**} |
| Blue collars | 2.67 | 2.42 | 0.24 |
| White collars | 0.44 | 0.37 | 0.06 |
| Apprentices | 0.20 | 0.15 | 0.05^{***} |
| Manufacturing $(0/1)$ | 0.22 | 0.19 | 0.03^{***} |
| Construction $(0/1)$ | 0.43 | 0.44 | -0.01 |
| Trade $(0/1)$ | 0.17 | 0.20 | -0.03^{***} |
| Tourism $(0/1)$ | 0.09 | 0.05 | 0.04^{***} |
| Other services $(0/1)$ | 0.09 | 0.12 | -0.03^{***} |
| Tradeble $(0/1)$ | 0.51 | 0.49 | 0.02^{*} |
| Upstream $(0/1)$ | 0.17 | 0.22 | -0.04^{***} |
| No. observations | 49126 | 3191 | |

Note: column Treat refers to firms in the same local labor market (SLL) as the oil fields, and Ctrl to those outside. Column Diff reports the difference between the means in the other two columns. Sectoral dummies are created by aggregating 2-digit ATECO codes.

Table 14: Descriptive statistics. Sample: periphery, 1990-1995.

| | Ctrl | Treat | Diff |
|---------------------------------|-----------------------|--------|----------------|
| Average wage (euro) | 113294 | 120076 | -67.82^{***} |
| Average wage (euro) - trim 99.9 | 113020 | 120076 | -70.56^{***} |
| Average wage (euro) - trim 99 | 1117.12 | 119318 | -76.06^{***} |
| Employees | 4.06 | 2.99 | 1.07 |
| Employees - right trim 99.9 | 3.55 | 2.99 | 0.56^{**} |
| Employees - right trim 99 | 3.02 | 2.79 | 0.23^{*} |
| Blue collars | 2.90 | 2.42 | 0.48 |
| White collars | 0.51 | 0.37 | 0.14 |
| Apprentices | 0.22 | 0.15 | 0.07^{***} |
| Manufacturing $(0/1)$ | 0.21 | 0.19 | 0.02^{**} |
| Construction $(0/1)$ | 0.40 | 0.44 | -0.04^{***} |
| Trade $(0/1)$ | 0.19 | 0.20 | -0.01 |
| Tourism $(0/1)$ | 0.09 | 0.05 | 0.04^{***} |
| Other services $(0/1)$ | 0.11 | 0.12 | -0.01 |
| Tradeble $(0/1)$ | 0.52 | 0.49 | 0.03^{***} |
| Upstream(0/1) | 0.18 | 0.22 | -0.03^{***} |
| No. observations | 56777 | 3191 | |

Table 15: Descriptive statistics. Sample: mountain, 1990-1995.

| | Ctrl | Treat | Diff |
|---------------------------------|-----------------------|--------|---------------|
| Average wage (euro) | 121579 | 120076 | 15.03 |
| Average wage (euro) - trim 99.9 | 1211.11 | 120076 | 10.35 |
| Average wage (euro) - trim 99 | 119272 | 119318 | -0.46 |
| Employees | 4.50 | 2.99 | 1.51^{*} |
| Employees - right trim 99.9 | 4.14 | 2.99 | 1.16^{***} |
| Employees - right trim 99 | 3.28 | 2.79 | 0.49^{***} |
| Blue collars | 3.09 | 2.42 | 0.66^{**} |
| White collars | 0.93 | 0.37 | 0.56^{*} |
| Apprentices | 0.20 | 0.15 | 0.05^{***} |
| Manufacturing $(0/1)$ | 0.20 | 0.19 | 0.01 |
| Construction $(0/1)$ | 0.40 | 0.44 | -0.04^{***} |
| Trade $(0/1)$ | 0.21 | 0.20 | 0.00 |
| Tourism $(0/1)$ | 0.05 | 0.05 | -0.00 |
| Other services $(0/1)$ | 0.14 | 0.12 | 0.03^{***} |
| Tradeble $(0/1)$ | 0.49 | 0.49 | -0.00 |
| Upstream $(0/1)$ | 0.21 | 0.22 | -0.01 |
| No. observations | 35426 | 3191 | |

Note: column Treat refers to firms in the same local labor market (SLL) as the oil fields, and Ctrl to those outside. Column Diff reports the difference between the means in the other two columns. Sectoral dummies are created by aggregating 2-digit ATECO codes.

| Table 16: | Descriptive | statistics. | Sample: | no Melfi, | 1990-1995. |
|-----------|-------------|-------------|---------|-----------|------------|
|-----------|-------------|-------------|---------|-----------|------------|

| | Ctrl | Treat | Diff |
|---------------------------------|-----------------------|--------|---------------|
| Average wage (euro) | 122909 | 120076 | 28.33** |
| Average wage (euro) - trim 99.9 | 122443 | 120076 | 23.67^{**} |
| Average wage (euro) - trim 99 | 120642 | 119318 | 13.24 |
| Employees | 5.00 | 2.99 | 2.01^{*} |
| Employees - right trim 99.9 | 4.49 | 2.99 | 1.50^{***} |
| Employees - right trim 99 | 3.65 | 2.79 | 0.86^{***} |
| Blue collars | 3.37 | 2.42 | 0.94^{***} |
| White collars | 1.04 | 0.37 | 0.67^{*} |
| Apprentices | 0.22 | 0.15 | 0.07^{***} |
| Manufacturing $(0/1)$ | 0.22 | 0.19 | 0.03^{***} |
| Construction $(0/1)$ | 0.30 | 0.44 | -0.14^{***} |
| Trade $(0/1)$ | 0.25 | 0.20 | 0.04^{***} |
| Tourism $(0/1)$ | 0.07 | 0.05 | 0.01^{**} |
| Other services $(0/1)$ | 0.17 | 0.12 | 0.05^{***} |
| Tradeble $(0/1)$ | 0.57 | 0.49 | 0.08^{***} |
| Upstream $(0/1)$ | 0.23 | 0.22 | 0.01 |
| No. observations | 176882 | 3191 | |

Table 17: Descriptive statistics. Sample: all above, 1990-1995.

| | Ctrl | Treat | Diff |
|---------------------------------|--------|--------|----------------|
| Average wage (euro) | 112326 | 120076 | -77.50^{***} |
| Average wage (euro) - trim 99.9 | 111820 | 120076 | -82.56^{***} |
| Average wage (euro) - trim 99 | 110275 | 119318 | -90.43^{***} |
| Employees | 2.64 | 2.99 | -0.34^{***} |
| Employees - right trim 99.9 | 2.64 | 2.99 | -0.34^{***} |
| Employees - right trim 99 | 2.62 | 2.79 | -0.17^{*} |
| Blue collars | 2.13 | 2.42 | -0.29^{***} |
| White collars | 0.26 | 0.37 | -0.11^{***} |
| Apprentices | 0.20 | 0.15 | 0.05^{***} |
| Manufacturing $(0/1)$ | 0.21 | 0.19 | 0.02 |
| Construction $(0/1)$ | 0.51 | 0.44 | 0.08^{***} |
| Trade $(0/1)$ | 0.15 | 0.20 | -0.05^{***} |
| Tourism $(0/1)$ | 0.06 | 0.05 | 0.00 |
| Other services $(0/1)$ | 0.08 | 0.12 | -0.04^{***} |
| Tradeble $(0/1)$ | 0.44 | 0.49 | -0.05^{***} |
| Upstream $(0/1)$ | 0.16 | 0.22 | -0.06^{***} |
| No. observations | 8951 | 3191 | |

| Industry | Quartile |
|---|----------|
| 05 Mining of coal and lignite | 4 |
| 06 Extraction of crude petroleum and natural gas | 4 |
| 08 Other mining and quarrying | 4 |
| 09 Mining support service activities | 4 |
| 10 Manufacture of food products | 1 |
| 11 Manufacture of beverages | 1 |
| 12 Manufacture of tobacco products | 1 |
| 13 Manufacture of textiles | 1 |
| 14 Manufacture of wearing apparel | 1 |
| 16 Manufacture of leather and related products | 1 |
| 17 Manufacture of wood [] | 1 |
| 17 Manufacture of paper and paper products | 2 |
| 10 Manufacture of cole and refined netroleuro producto | 1 |
| 20 Manufacture of cloke and reined petroleum products | 4 |
| 20 Manufacture of chemicals and chemical products | 4 |
| 21 Manufacture of basic pharmaceutical products [] | 3 |
| 22 Manufacture of rubber and plastic products | 3 |
| 23 Manufacture of other non-metallic mineral products | 4 |
| 24 Manufacture of basic metals | 2 |
| 25 Manufacture of fabricated metal products [] | 1 |
| 26 Manufacture of computer, eletronic and optical products | 2 |
| 27 Manufacture of electrical equipment | 3 |
| 28 Manufacture of machinery and equipment n.e.c | 4 |
| 29 Manufacture of motor vehicles, trailers and semi-trailers | 4 |
| 30 Manufacture of other transport equipment | 4 |
| 22 Other menufacture | 1 |
| 32 Other manufacture | 1 |
| 25 Electricity and instantion of machinery and equipment | 4 |
| 26 Water collection, treatment and dispessed estivities [| 4 |
| 27 Server conection, treatment and disposal activities [] | 4 |
| 28 Wests collection treatment and dispaced activities [| 4 |
| 20 Demodiation activities and other waste management [| 4 |
| 41 Construction of buildings | 4 |
| 42 Civil onginogring | 1 |
| 42 Civil engineering 42 Specialized construction activities | 1 |
| 45 Wholesele and rotail traind and renain of motor vahiales [1] | 1 |
| 45 Wholesale and retail traind and repair of motor vehicles [] | 4 |
| 40 Wholesale trade, except of motor vehicles and motorcycles | 4 |
| 40 Land transport and transport via pipelines | 4 |
| 50 Water transport | 4 |
| 50 Water transport | 4 |
| 52 Warehousing and support activities for transportation | 4 |
| 52 Postal and courier activities | |
| 55 Accomposition | 4 |
| 56 Food and hoverage service activities | 3 |
| 58 Publishing activities | 3 |
| 50 Motion picture, video and television programme [] | 3 |
| 60 Programming and broadcasting activities | 3 |
| 61 Telecommunications | 3 |
| 62 Computer programming consultancy and related [] | 4 |
| 63 Information service activities | 4 |
| 64 Financial corvice activities accort incurance | 4 |
| 65 Insurance reinsurance and pension funding [] | 3 |
| 66 Activities auxiliary to financial services [] | 1 |
| 68 Real estate activities | 1 |
| 60 Legal and accounting activities | 2 |
| 70 Activities of head offices: management consultancy [] | 2 2 |
| 71 Architectural and engineering activities [1] | 5 / |
| 72 Scientific research and development | ± Δ |
| 73 Advertising and market research | 4 2 |
| 74 Other professional scientific and technical activities | 5 / |
| 75 Veterinary activities | -± Λ |
| is reconnery activities | -1 |

Table 18: Industries by quartile of upstreamness.

| Table | 18 | _ | continued | from | previous | a page |
|-------------------|-------------|-----------|---------------------|---------------|----------|----------|
| | | | Industry | | | Quartile |
| 77 Rent | al and le | asing ac | tivities | | | 4 |
| $78 \mathrm{Emp}$ | loyment | activitie | es | | | 1 |
| 79 Trav | el agency | , tour o | perator reservatio | on service [. |] | 3 |
| 80 Secu | rity and i | investig | ation activities | | | 3 |
| 81 Servi | ices to bu | uldings | and landscape act | tivities | | 3 |
| 82 Offic | e adminis | strative | , office support an | d other [|] | 3 |
| 90 Crea | tive, arts | and en | tertainment activi | ities | | 3 |
| 91 Libra | aries, arcl | hives, m | useums and other | r [] | | 3 |
| $92~\mathrm{Gam}$ | bling and | l bettin | g activities | | | 3 |
| 93 Spor | ts activit | ies and | amusement and r | ecreation [. |] | 4 |
| 94 Activ | vities of r | nember | ship organisations | 5 | | 4 |
| 95 Repa | air of com | puters | [] | | | 3 |
| 96 Othe | er persona | al servic | e activities | | | 2 |

- N. 1361 How do firms adjust to a negative labor supply shock? Evidence form migration outflows, by Emanuele Dicarlo (February 2022).
- N. 1362 *Nowcasting the state of the Italian economy: the role of financial markets*, by Donato Ceci and Andrea Silvestrini (February 2022).
- N. 1363 *Insurers' investments before and after the Covid-19 outbreak*, by Federico Apicella, Raffaele Gallo and Giovanni Guazzarotti (February 2022).
- N. 1364 *Making subsidies work: rules vs. discretion*, by Federico Cingano, Paolo Pinotti, Filippo Palomba and Enrico Rettore (March 2022).
- N. 1365 Foreign monetary policy and domestic inflation in emerging markets, by Marco Flaccadoro and Valerio Nispi Landi (April 2022).
- N. 1366 *Monetary policy in the open economy with digital currencies*, by Pietro Cova, Alessandro Notarpietro, Patrizio Pagano and Massimiliano Pisani (April 2022).
- N. 1367 The role of non-bank financial institutions in the intermediation of capital flows to emerging markets, by Alessandro Moro and Alessandro Schiavone (April 2022).
- N. 1368 Exchange rate pass-through in small, open, commodity-exporting economies: lessons from Canada, by Marco Flaccadoro (April 2022).
- N. 1369 Public guarantees and credit additionality during the Covid-19 pandemic, by Giuseppe Cascarino, Raffaele Gallo, Francesco Palazzo and Enrico Sette (April 2022).
- N.1370 The effects of local demand and supply restrictions on markup, by Antonio Acconcia and Elisa Scarinzi (June 2022).
- N. 1371 Mutual fund trading and ESG stock resilience during the Covid-19 stock market crash, by Rui Albuquerque, Yrjö Koskinen and Raffaele Santioni (June 2022).
- N. 1372 *Higher capital requirements and credit supply: evidence from Italy*, by Maddalena Galardo and Valerio Vacca (June 2022).
- N. 1373 Voluntary support and ring-fencing in cross-border banks, by Gyoengyi Loranth, Anatoli Segura and Jing Zeng (June 2022).
- N. 1374 Plt ain't where you're from it's where you're at: firm effects, state dependence, and the gender wage gap, by Sabrina Di Addario, Patrick Kline, Raffaele Saggio and Mikkel Søelvsten (June 2022).
- N. 1375 New facts on consumer price rigidity in the Euro Area, by Erwan Gautier, Cristina Conflitti, Riemer P. Faber, Brian Fabo, Ludmila Fadejeva, Valentin Jouvanceau, Jan-Oliver Menz, Teresa Messner, Pavlos Petroulas, Pau Roldan-Blanco, Fabio Rumler, Sergio Santork, Elisabeth Wieland and Hélène Zimmer (July 2022).
- N. 1376 Connecting to power: political connections, innovation, and firm dynamics, by Ufuk Akcigit, Salomé Baslandze and Francesca Lotti (July 2022).
- N. 1377 "Green" fiscal policy measures and non-standard monetary policy in the *Euro Area*, by Anna Bartocci, Alessandro Notarpietro and Massimiliano Pisani (July 2022).
- N.1378 *Firm liquidity and the transmission of monetary policy*, by Margherita Bottero and Stefano Schiaffi (July 2022).
- N. 1379 *The impact of Covid-19 on the european short-term rental market*, by Elisa Guglielminetti, Michele Loberto and Alessandro Mistretta (July 2022).
- N.1380 An analysis of objective inflation expectations and inflation risk premia, by Sara Cecchetti, Adriana Grasso and Marcello Pericoli (July 2022).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2020

- ALESSANDRI P. and M. BOTTERO, *Bank lending in uncertain times,* R European Economic Review, V. 128, WP 1109 (April 2017).
- ANTUNES A. and V. ERCOLANI, Public debt expansions and the dynamics of the household borrowing constraint, Review of Economic Dynamics, v. 37, pp. 1-32, WP 1268 (March 2020).
- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities*, Journal of Business & Economic Statistics, v. 38, 4, pp. 826-838, **WP 974 (October 2014)**.
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, Discretion and supplier selection in public procurement, Journal of Law, Economics, and Organization, v. 37, 1, pp. 134-166, WP 1178 (June 2018)
- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, v. 16, 6, pp. 137-184, **WP 1201 (November 2018).**
- BOTTERO M., F. MEZZANOTTI and S. LENZU, Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy, Journal of International Economics, v. 126, article 103328, WP 1032 (October 2015).
- BRIPI F., D. LOSCHIAVO and D. REVELLI, Services trade and credit frictions: evidence with matched bank *firm data*, The World Economy, v. 43, 5, pp. 1216-1252, **WP 1110 (April 2017).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, *Venture capitalists at work: a Diff-in-Diff approach at late-stages of the screening process*, Journal of Business Venturing, v. 35, 3, **WP 1131 (September 2017).**
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome,* Journal of Regional Science, v. 60, 4, pp. 801-822, WP 1208 (February 2019).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis,* Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, **WP 1179 (June 2018).**
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, v. 102, **WP 1152 (December 2017).**
- D'ALESSIO G., *Measurement errors in survey data and the estimation of poverty and inequality indices,* Statistica Applicata - Italian Journal of Applied Statistics, v. 32, 3, **WP 1116 (June 2017).**
- DE BLASIO G., I. DE ANGELIS and L. RIZZICA, *Lost in corruption. Evidence from EU funding to Southern Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 6, 3, pp. 355–377, WP 1180 (December 2016).
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises,* Review of World Economic, v. 156, 4, pp. 859 885, WP 1217 (April 2019).
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- ERCOLANI V. and F. NATOLI, *Forecasting US recessions: the role of economic uncertainty*, Economics Letters, v. 193, WP 1299 (October 2020).
- MAKINEN T., L. SARNO and G. ZINNA, *Risky bank guarantees*, Journal of Financial Economics, v. 136, 2, pp. 490-522, **WP 1232 (July 2019).**
- MODENA F., E. RETTORE and G. M. TANZI, *The effect of grants on university dropout rates: evidence from the Italian case*, Journal of Human Capital, v. 14, 3, pp. 343-370, WP 1193 (September 2018).
- NISPI LANDI V., *Capital controls spillovers*, Journal of International Money and Finance, v. 109, WP 1184 (July 2018).
- PERICOLI M., On risk factors of the stock-bond correlation, International Finance, v. 23, 3, pp. 392-416, WP 1198 (November 2018).
- PIETRUNTI M. and F. M. SIGNORETTI, Unconventional monetary policy and household debt: the role of cashflow effects, Journal of Macroeconomics, v. 64, Article 103201, WP 1142 (October 2017).
- RAINONE E., *The network nature of OTC interest rates*, Journal of Financial Markets, v.47, article 100525, WP 1022 (July 2015).
- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, **WP 1059 (March 2016).**
- RIZZICA L., *Raising aspirations and higher education. Evidence from the UK's widening participation policy,* Journal of Labor Economics, v. 38, 1, pp. 183-214, **WP 1188 (September 2018).**
- RIZZICA L. and M. TONELLO, Persuadable perceptions: the effect of media content on beliefs about

"TEMI" LATER PUBLISHED ELSEWHERE

corruption, Economic Policy, v. 35, 104, pp. 679-737, WP 1043 (November 2016).

- SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, v. 24, 4, pp. 773-811, WP 1146 (October 2017).
- SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, *Bank Quality, judicial efficiency and loan repayment delays in Italy,* Journal of Finance, v. 75, 4, pp. 2139-2178, **WP 1072 (July 2016).**

2021

- ACCETTURO A., A. LAMORGESE, S. MOCETTI and D. PELLEGRINO, *Housing Price elasticity and growth: evidence from Italian cities,* Journal of Economic Geography, v. 21, 3, pp. 367-396, WP 1267 (March 2020).
- AFFINITO M. and M. PIAZZA, Always look on the bright side? Central counterparties and interbank markets during the financial crisis, International Journal of Central Banking, v. 17, 1, pp. 231-283, WP 1181 (July 2018).
- ALBANESE G., E. CIANI and G. DE BLASIO, *Anything new in town? The local effects of urban regeneration policies in Italy*, Regional Science and Urban Economics, v. 86, **WP 1214 (April 2019)**.
- ALBANESE G., G. DE BLASIO and A. LOCATELLI, *Does EU regional policy promote local TFP growth? Evidence from the Italian Mezzogiorno*, Papers in Regional Science, v. 100, 2, pp. 327-348, **WP 1253 (December 2019).**
- ALBERTAZZI A., A. NOBILI and F. M. SIGNORETTI, *The bank lending channel of conventional and unconventional monetary policy*, Journal of Money, Credit and Banking, v. 53, 2-3, pp. 261-299, **WP 1094 (Jan 2017)**.
- ANZUINI A. and L. ROSSI, Fiscal policy in the US: a new measure of uncertainty and its effects on the American economy, Empirical Economics, v. 61, 6, pp. 2613-2634, WP 1197 (November 2018).
- APRIGLIANO V. and D. LIBERATI, Using credit variables to date business cycle and to estimate the probabilities of recession in real time, The Manchester School, v. 89, 51, pp. 76-96, WP 1229 (July 2019).
- AUER S., M. BERNARDINI and M. CECIONI, *Corporate leverage and monetary policy effectiveness in the euro area,* European Economic Review, v. 140, Article 103943, **WP 1258 (December 2019).**
- BANERJEE R, L. GAMBACORTA and E. SETTE, *The real effects of relationship lending*, Journal of Financial Intermediation, v. 48, Article 100923, **WP 1133 (September 2017).**
- BARONE G., F. DAVID, G. DE BLASIO and S. MOCETTI, *How do house prices respond to mortgage supply?*, Journal of Economic Geography, v. 21, 1, pp.127-140, **WP 1282 (June 2020).**
- BARONE G. and S. MOCETTI, Intergenerational mobility in the very long run: Florence 1427-2011, Review of Economic Studies, v. 88, 4, pp. 1863–1891, WP 1060 (April 2016).BARTOCCI A., L. BURLON, A. NOTARPIETRO and M. PISANI, Macroeconomic effects of non-standard monetary policy measures in the Euro Area: the role of corporate bond purchases, The Manchester School, v. 89, S1, pp. 97-130, WP 1241 (Oct 2019).
- BATINI N., A. CANTELMO, G. MELINA and S. VILLA, *How loose, how tight? A measure of monetary and fiscal stance for the euro area,* Oxford Economic Papers, v. 73, 4, pp. 1536-1556, **WP 1295 (September 2020).**
- BENETTON M. and D. FANTINO, *Targeted monetary policy and bank lending behavior*, Journal of Financial Economics, v. 142, 1, pp. 404-429, **WP 1187 (September 2018).**
- BUSETTI F., M. CAIVANO and D. DELLE MONACHE, Domestic and global determinants of inflation: evidence from expectile regression, Oxford Bulletin of Economics and Statistics, v. 83, 4, pp. 982-1001, WP 1225 (June 2019).
- BUSETTI F., M. CAIVANO, D. DELLE MONACHE and C. PACELLA, *The time-varying risk of Italian GDP*, Economic Modelling, v. 101, Article 105522, WP 1288 (July 2020).
- BUSETTI F., S. NERI, A. NOTARPIETRO and M. PISANI, Monetary Policy strategies in the new normal: a modelbased analysis for the Euro Area, Journal of Macroeconomics, v. 70, Article 103366, WP 1308 (December 2020).
- CAPOLONGO A. and C. PACELLA, Forecasting inflation in the Euro Area: countries matter, Empirical Economics, v. 61, 4, pp. 2477-2499, WP 1224 (June 2019).
- CARMIGNANI A., G. DE BLASIO, C. DEMMA and A. D'IGNAZIO, *Urbanization and firm access to credit,* Journal of Regional Science, v. 61, 3, pp. 597-622, WP 1222 (June 2019).
- CORNELI F., *Financial integration without financial development*, Atlantic Economic Journal, v. 49, 2, pp. 201-220, **WP 1120 (June 2017).**
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, Secular stagnation, R&D, public investment and monetary policy: a global-model perspective, Macroeconomic Dynamics, v. 25, 5, pp. 1267-1287, WP 1156 (December 2017).

- DE PHILIPPIS M., *Multitask agents and incentives: the case of teaching and research for university professors,* Economic Journal, v. 131, 636, pp. 1643-1681, **WP 1042 (December 2015).**
- DEL PRETE S. and M. L. STEFANI, Women as "Gold Dust": gender diversity in top boards and the performance of Italian banks, Economic Notes, Monte dei Paschi di Siena, v. 50, 2, e12183, WP 1014 (June 2015).
- FERRERO G., M. LOBERTO and M. MICCOLI, *The assets' pledgeability channel of unconventional monetary policy*, Economic Inquiry, v. 59, 4, pp. 1547-1568, **WP 1119 (June 2017).**
- FIDORA M., C. GIORDANO and M. SCHMITZ, *Real exchange rate misalignments in the Euro Area,* Open Economies Review, v. 32, 1, pp. 71-107, **WP 1162 (January 2018).**
- GAMBACORTA L., G. RICOTTI, S. SUNDARESAN and Z. WANG, *Tax effects on bank liability structure*, European Economic Review, v. 138, Article 103820, **WP 1101 (February 2017).**
- HERTWECK M., V. LEWIS and S. VILLA, *Going the extra mile: effort by workers and job-seekers,* Journal of Money, Credit and Banking, v. 54, 8, pp. 2099-2127, **WP 1277 (June 2020).**
- LI F., A. MERCATANTI, T. MAKINEN and A. SILVESTRINI, *A regression discontinuity design for ordinal running variables: evaluating central bank purchases of corporate bonds,* The Annals of Applied Statistics, v. 15, 1, pp. 304-322, **WP 1213 (March 2019).**
- LOSCHIAVO D., *Big-city life (dis)satisfaction? The effect of urban living on subjective well-being*, Journal of Economic Behavior & Organization, vol. 192, pp. 740-764, **WP 1221 (June 2019).**
- LOSCHIAVO D., *Household debt and income inequality: evidence from Italian survey data*, Review of Income and Wealth. v. 67, 1, pp. 61-103, **WP 1095 (January 2017).**
- METELLI L. and F. NATOLI, *The international transmission of US tax shocks: a proxy-SVAR approach*, IMF Economic Review, v. 69, 2, pp. 325-356, **WP 1223 (June 2019).**
- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, v. 32, 1, pp. 183-211, WP 1200 (November 2018).
- PAPETTI A., *Demographics and the natural real interest rate: historical and projected paths for the Euro Area*, Journal of Economic Dynamics and Control, v. 132, Article 04209, WP 1306 (November 2020).
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, v. 39, 2, pp. 575-588, WP 1092 (January 2017).

2022

- ANDINI M., M. BOLDRINI, E. CIANI, G. DE BLASIO, A. D'IGNAZIO and A. PALADINI, Machine learning in the service of policy targeting: the case of public credit guarantees, Journal of Economic Behavior & Organization, v. 198, pp. 434-475, WP 1206 (February 2019).
- ANGELICO C., J. MARCUCCI, M. MICCOLI and F. QUARTA, Can we measure inflation expectations using twitter?, Journal of Econometrics, v. 228, 2, pp. 259-277, WP 1318 (February 2021).
- BOTTERO M., C. MINOIU, J. PEYDRÒ, A. POLO, A. PRESBITERO and E. SETTE, *Expansionary yet different: credit supply and real effects of negative interest rate policy,* Journal of Financial Economics, v. 146, 2, pp. 754-778, **WP 1269 (March 2020).**
- CANTELMO A., *Rare disasters, the natural interest rate and monetary policy,* Oxford Bulletin of Economics and Statistics, v. 84, 3, pp. 473-496, **WP 1309 (December 2020).**
- CARRIERO A., F. CORSELLO and M. MARCELLINO, *The global component of inflation volatility*, Journal of Applied Econometrics, v. 37, 4, pp. 700-721, **WP 1170 (May 2018).**
- CIAPANNA E. and G. ROVIGATTI, *The grocery trolley race in times of Covid-19. Evidence from Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 8, 2, pp. 471-498, **WP 1341 (June 2021)**.
- FAIELLA I. and A. MISTRETTA, The net zero challenge for firms' competitiveness, Environmental & Resource Economics, v. 83, pp. 85-113, WP 1259 (February 2020).
- GUISO L., A. POZZI, A. TSOY, L. GAMBACORTA and P. E. MISTRULLI, *The cost of steering in financial markets:* evidence from the mortgage market, Journal of Financial Economics, v.143, 3, pp. 1209-1226, **WP 1252 (December 2019).**
- LAMORGESE A. and D. PELLEGRINO, *Loss aversion in housing appraisal: evidence from Italian homeowners,* Journal of Housing Economics, v. 56, Article 101826, WP 1248 (November 2019).
- LI F., T. MÄKINEN, A. MERCATANTI and A. SILVESTRINI, *Causal analysis of central bank holdings of corporate bonds under interference,* Economic Modelling, v.113, Article 105873, WP 1300 (November 2020).
- MIRENDA L., M. SAURO and L. RIZZICA, *The economic effects of mafia: firm level evidence*, American Economic Review, vol. 112, 8, pp. 2748-2773, **WP 1235 (October 2019).**
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility,* Journal of Human Resources, v. 57, 2, pp. 525-554, WP 1182 (July 2018).

- ROSSI P. and D. SCALISE, *Financial development and growth in European regions*, Journal of Regional Science, v. 62, 2, pp. 389-411, WP 1246 (November 2019).
- SCHIVARDI F., E. SETTE and G. TABELLINI, *Credit misallocation during the European financial crisis,* Economic Journal, v. 132, 641, pp. 391-423, **WP 1139 (September 2017).**
- TABOGA M., Cross-country differences in the size of venture capital financing rounds: a machine learning approach, Empirical Economics, v. 62, 3, pp. 991-1012, WP 1243 (November 2019).

FORTHCOMING

- APRIGLIANO V., S. EMILIOZZI, G. GUAITOLI, A. LUCIANI, J. MARCUCCI and L. MONTEFORTE, *The power of textbased indicators in forecasting Italian economic activity*, International Journal of Forecasting, WP 1321 (March 2021).
- BRONZINI R., A. D'IGNAZIO and D. REVELLI, *Financial structure and bank relationships of Italian multinational firms*, Journal of Multinational Financial Management, **WP 1326 (March 2021).**
- FERRARI A. and V. NISPI LANDI, *Toward a green economy: the role of central bank's asset purchases,* International Journal of Central Banking, **WP 1358 (February 2022).**
- LILLA F., Volatility bursts: a discrete-time option model with multiple volatility components, Journal of Financial Econometrics, WP 1336 (June 2021).
- LOBERTO M, Foreclosures and house prices, Italian Economic Journal / Rivista italiana degli economisti, WP 1325 (March 2021).
- LOBERTO M, A. LUCIANI and M. PANGALLO, *What do online listings tell us about the housing market?*, International Journal of Central Banking, **WP 1171 (April 2018).**
- PERICOLI M. and M. TABOGA, *Nearly exact Bayesian estimation of non-linear no-arbitrage term-structure models*, Journal of Financial Econometrics, **WP 1189 (September 2018).**
- RIZZICA L., G. ROMA and G. ROVIGATTI, *The effects of shop opening hours deregulation: evidence from Italy*, The Journal of Law and Economics, **WP 1281 (June 2020).**
- TANZI G. M., Scars of youth non-employment and labour market conditions, Italian Economic Journal / Rivista italiana degli economisti, **WP 1312 (December 2020).**